**EPJ manuscript No.** (will be inserted by the editor)

CERN-PH-EP-2011-114

Submitted to Eur. Phys. J. C

# Performance of Missing Transverse Momentum Reconstruction in Proton-Proton Collisions at $\sqrt{s}$ = 7 TeV with ATLAS

The ATLAS Collaboration

CERN, 1211 Geneva 23, Switzerland

December 6, 2011

**Abstract.** The measurement of missing transverse momentum in the ATLAS detector, described in this paper, makes use of the full event reconstruction and a calibration based on reconstructed physics objects. The performance of the missing transverse momentum reconstruction is evaluated using data collected in pp collisions at a centre-of-mass energy of 7 TeV in 2010. Minimum bias events and events with jets of hadrons are used from data samples corresponding to an integrated luminosity of about  $0.3 \text{ nb}^{-1}$  and  $600 \text{ nb}^{-1}$  respectively, together with events containing a Z boson decaying to two leptons (electrons or muons) or a W boson decaying to a lepton (electron or muon) and a neutrino, from a data sample corresponding to an integrated luminosity of about 36 pb<sup>-1</sup>. An estimate of the systematic uncertainty on the missing transverse momentum scale is presented.

#### 1 Introduction

In a collider event the missing transverse momentum is defined as the momentum imbalance in the plane transverse to the beam axis, where momentum conservation is expected. Such an imbalance may signal the presence of unseen particles, such as neutrinos or stable, weakly-interacting supersymmetric (SUSY) particles. The vector momentum imbalance in the transverse plane is obtained from the negative vector sum of the momenta of all particles detected in a *pp* collison and is denoted as missing transverse momentum,  $E_{T}^{miss}$ . The symbol  $E_{T}^{miss}$  is used for its magnitude.

A precise measurement of the missing transverse momentum,  $E_T^{\text{miss}}$ , is essential for physics at the LHC. A large  $E_T^{\text{miss}}$  is a key signature for searches for new physics processes such as SUSY and extra dimensions. The measurement of  $E_T^{\text{miss}}$  also has a direct impact on the quality of a number of measurements of Standard Model (SM) physics, such as the reconstruction of the top-quark mass in  $t\bar{t}$  events. Furthermore, it is crucial in the search for the Higgs boson in the decay channels  $H \to WW$ and  $H \to \tau \tau$ , where a good  $E_T^{\text{miss}}$  measurement improves the reconstruction of the Higgs boson mass [1].

This paper describes an optimized reconstruction and calibration of  $E_{\rm T}^{\rm miss}$  developed by the ATLAS Collaboration. The performance achieved represents a significant improvement compared to earlier results [2] presented by ATLAS. The optimal reconstruction of  $E_{\rm T}^{\rm miss}$  in the ATLAS detector is complex and validation with data, in terms of resolution, scale and tails, is essential. A number of data samples encompassing a variety of event topologies are used. Specifically, the event samples used to assess the quality of the  $E_{\rm T}^{\rm miss}$  reconstruction are: min-

imum bias events, events where jets at high transverse momentum are produced via strong interactions described by Quantum Chromodynamics (QCD) and events with leptonically decaying W and Z bosons. This allows the full exploitation of the detector capability in the reconstruction and calibration of different physics objects and optimization of the  $E_T^{miss}$  calculation. Moreover, in events with  $W \rightarrow \ell v$ , where  $\ell$  is an electron or muon, the  $E_T^{miss}$  performance can be studied in events where genuine  $E_T^{miss}$  is present due to the neutrino, thus allowing a validation of the  $E_T^{miss}$  scale. In simulated events, the genuine  $E_T^{miss}$ ,  $E_T^{miss,True}$ , is calculated from all generated noninteracting particles in the event and it is also referred to as true  $E_T^{miss}$  in the following.

An important requirement on the measurement of  $E_{\rm T}^{\rm miss}$  is the minimization of the impact of limited detector coverage, finite detector resolution, the presence of dead regions and different sources of noise that can produce fake  $E_{\rm T}^{\rm miss}$ . The AT-LAS calorimeter coverage extends to large pseudorapidities <sup>1</sup> to minimize the impact of high energy particles escaping in the very forward direction. Even so, there are inactive transition regions between different calorimeters that produce fake  $E_{\rm T}^{\rm miss}$ . Dead and noisy readout channels in the detector, if present, as

brought to you by TCORE

<sup>&</sup>lt;sup>1</sup> ATLAS uses a right-handed coordinate system with its origin at the nominal interaction point (IP) in the centre of the detector and the *z*-axis coinciding with the axis of the beam pipe. The *x*-axis points from the IP to the centre of the LHC ring, and the *y* axis points upward. Cylindrical coordinates  $(r, \phi)$  are used in the transverse plane,  $\phi$ being the azimuthal angle around the beam pipe. The pseudorapidity is defined in terms of the polar angle  $\theta$  as  $\eta = -\ln \tan(\theta/2)$ .

well as cosmic-ray and beam-halo muons crossing the detector, will produce fake  $E_T^{\text{miss}}$ . Such sources can significantly enhance the background from multi-jet events in SUSY searches with large  $E_T^{\text{miss}}$  or the background from  $Z \rightarrow \ell \ell$  events accompanied by jets of high transverse momentum ( $p_T$ ) in Higgs boson searches in final states with two leptons and  $E_T^{\text{miss}}$ . Cuts are applied to clean the data against all these sources (see Section 3), and more severe cuts to suppress fake  $E_T^{\text{miss}}$  are applied in analyses for SUSY searches, after which, for selected events with high- $p_T$  jets, the tails of the  $E_T^{\text{miss}}$  distributions are well described by MC simulation [3].

This paper is organised as follows. Section 2 gives a brief introduction to the ATLAS detector. Section 3 and Section 4 describe the data and Monte Carlo samples used and the event selections applied. Section 5 outlines how  $E_T^{miss}$  is reconstructed and calibrated. Section 6 presents the  $E_T^{miss}$  performance for data and Monte Carlo simulation, first in minimum bias and jet events and then in Z and W events. The systematic uncertainty on the  $E_T^{miss}$  absolute scale is discussed in Section 7. Section 8 describes the determination of the  $E_T^{miss}$  scale in-situ using  $W \rightarrow \ell v$  events. Finally, the conclusions are given in Section 9.

#### 2 The ATLAS Detector

The ATLAS detector [1] is a multipurpose particle physics apparatus with a forward-backward symmetric cylindrical geometry and near  $4\pi$  coverage in solid angle. The inner tracking detector (ID) covers the pseudorapidity range  $|\eta| < 2.5$ , and consists of a silicon pixel detector, a silicon microstrip detector (SCT), and, for  $|\eta| < 2.0$ , a transition radiation tracker (TRT). The ID is surrounded by a thin superconducting solenoid providing a 2 T magnetic field. A high-granularity lead/liquidargon (LAr) sampling electromagnetic calorimeter covers the region  $|\eta| < 3.2$ . An iron/scintillator-tile calorimeter provides hadronic coverage in the range  $|\eta| < 1.7$ . LAr technology is also used for the hadronic calorimeters in the end-cap region  $1.5 < |\eta| < 3.2$  and for both electromagnetic and hadronic measurements in the forward region up to  $|\eta| < 4.9$ . The muon spectrometer surrounds the calorimeters. It consists of three large air-core superconducting toroid systems, precision tracking chambers providing accurate muon tracking out to  $|\eta| = 2.7$ , and additional detectors for triggering in the region  $|\eta| < 2.4$ .

#### 3 Data samples and event selection

During 2010 a large number of proton-proton collisions, at a centre-of-mass energy of 7 TeV, were recorded with stable proton beams as well as nominal magnetic field conditions. Only data with a fully functioning calorimeter, inner detector and muon spectrometer are analysed.

Cuts are applied to clean the data sample against instrumental noise and out-of-time energy deposits in the calorimeter (from cosmic-rays or beam-induced background). Topological clusters reconstructed in the calorimeters (see Section 5.1) at the electromagnetic energy (EM) scale<sup>2</sup> are used as the inputs to the jet finder [4]. In this paper the anti- $k_t$  algorithm [5], with distance parameter R = 0.6, is used for jet reconstruction. The reconstructed jets are required to pass basic jetquality selection criteria. In particular events are rejected if any jet in the event with transverse momentum  $p_T>20$  GeV is caused by sporadic noise bursts in the end-cap region, coherent noise in the electromagnetic calorimeter or reconstructed from large out-of-time energy deposits in the calorimeter. These cuts largely suppress the residual sources of fake  $E_T^{miss}$  due to those instrumental effects which remain after the data-quality requirements.

The 2010 data sets used in this paper correspond to a total integrated luminosity [6,7] of approximately 600 nb<sup>-1</sup> for jet events and to 0.3 nb<sup>-1</sup> for minimum bias events. Trigger and selection criteria for these events are described in Section 3.1. For the  $Z \rightarrow \ell \ell$  and  $W \rightarrow \ell \nu$  channels, the samples analysed correspond to an integrated luminosity of approximately 36 pb<sup>-1</sup>. Trigger and selection criteria, similar to those developed for the W/Z cross-section measurement [8], are applied. These criteria are described in Sections 3.2 and 3.3.

#### 3.1 Minimum bias and di-jet event selection

For the minimum bias events, only the early period of data taking, with a minimal pile-up contribution, is studied. Selected minimum bias events were triggered by the minimum bias trigger scintillators (MBTS), mounted at each end of the detector in front of the LAr end-cap calorimeter cryostats [9].

Events in the QCD jet sample are required to have passed the first-level calorimeter trigger, which indicates a significant energy deposit in a certain region of the calorimeter, with the most inclusive trigger with a nominal  $p_{\rm T}$  threshold at 15 GeV. The event sample used in this analysis consists of two subsets of 300 nb<sup>-1</sup> each, corresponding to two periods with different pile-up and trigger conditions<sup>3</sup>. One subset corresponds to the periods with lower pile-up conditions with, on average, 1 to 1.6 reconstructed vertices per event. The other subset corresponds to the periods with higher pileup conditions, where the peak number of visible inelastic interactions per bunch crossing goes up to 3. In the following, di-jet events are used, selected requiring the presence of exactly two jets with  $p_{\rm T} > 25$  GeV and  $|\eta| < 4.5$ . Jets are calibrated with the local hadronic calibration (see Section 5.1).

For each event, at least one good primary vertex is required with a z displacement from the nominal pp interaction point less than 200 mm and with at least five associated tracks. After selection, the samples used in the analysis presented here correspond to 14 million minimum bias events and 13 million di-jet events.

#### 3.2 $Z \rightarrow \ell \ell$ event selection

Candidate  $Z \rightarrow \ell \ell$  events, where  $\ell$  is an electron or a muon, are required to pass an  $e/\gamma$  or muon trigger with a  $p_T$  threshold between 10 and 15 GeV, where the exact trigger selection varies

<sup>&</sup>lt;sup>2</sup> The EM scale is the basic calorimeter signal scale for the AT-LAS calorimeters. It provides the correct scale for energy deposited

by electromagnetic showers. It does not correct for the lower energy hadron shower response nor for energy losses in the dead material.

<sup>&</sup>lt;sup>3</sup> Pile-up in the following refers to the contribution of additional pp collisions superimposed on the hard physics process.

depending on the data period analysed. For each event, at least one good primary vertex, as defined above, is required.

The selection of  $Z \rightarrow \mu\mu$  events requires the presence of exactly two good muons. A good muon is defined to be a muon reconstructed in the muon spectrometer with a matched track in the inner detector with transverse momentum above 20 GeV and  $|\eta| < 2.5$  [10]. Additional requirements on the number of hits used to reconstruct the track in the inner detector are applied. The *z* displacement of the muon track from the primary vertex is required to be less than 10 mm. Isolation cuts are applied around the muon track.

The selection of  $Z \rightarrow ee$  events requires the presence of exactly two identified electrons with  $|\eta| < 2.47$ , which pass the "medium" identification criteria [8, 11] and have transverse momenta above 20 GeV. Electron candidates in the electromagnetic calorimeter transition region,  $1.37 < |\eta| < 1.52$ , are not considered for this study. Additional cuts are applied to remove electrons falling into regions where the readout of the calorimeter was not fully operational.

In both the  $Z \rightarrow ee$  and the  $Z \rightarrow \mu\mu$  selections, the two leptons are required to have opposite charge and the reconstructed invariant mass of the di-lepton system,  $m_{\ell\ell}$ , is required to be consistent with the Z mass,  $66 < m_{\ell\ell} < 116$  GeV.

With these selection criteria, about 9000  $Z \rightarrow ee$  and 13000  $Z \rightarrow \mu\mu$  events are selected. The estimated background contribution to these samples is less than 2% in both channels [8].

#### 3.3 $W \rightarrow \ell v$ event selection

Lepton candidates are selected with lepton identification criteria similar to those used for the *Z* analysis. The differences for the selection of  $W \rightarrow ev$  events are that the "tight" electron identification criteria [11,8] are used and an isolation cut is applied on the electron cluster in the calorimeter to reduce contamination from QCD jet background. The event is rejected if it contains more than one reconstructed lepton. The  $E_T^{miss}$ , calculated as described in Section 5, is required to be greater than 25 GeV, and the reconstructed lepton- $E_T^{miss}$  transverse mass,  $m_T$ , is required to be greater than 50 GeV.

With these selection criteria, about  $8.5 \times 10^4 W \rightarrow ev$  and  $1.05 \times 10^5 W \rightarrow \mu v$  events are selected. The background contribution to these samples is estimated to be about 5% in both channels [8].

#### 4 Monte Carlo simulation samples

Monte Carlo (MC) events are generated using the PYTHIA6 program [12] with the ATLAS minimum bias tune (AMBT1) of the PYTHIA fragmentation and hadronisation parameters [13]. The generated events are processed with the detailed GEANT4 [14] simulation of the ATLAS detector.

The minimum bias MC event samples are generated using non-diffractive as well as single- and double-diffractive processes, where the different components are weighted according to the cross-sections given by the event generators.

The jet MC samples, generated using a 2-to-2 QCD matrix element and subsequent parton shower development, are used for comparison with the two subsets of data taken with different pile-up conditions. In the earlier sample the fraction of events with at least two observed interactions is at most of the order of 8 - 10 %, while in the sample taken later in 2010 this fraction ranges from 10 % to more than 50 %. These samples are generated in the  $p_{\rm T}$  range 8 - 560 GeV, in separated parton  $p_{\rm T}$  bins to provide a larger statistics also in the high- $p_{\rm T}$  bins. Each sample is weighted according to its cross-section.

MC events for the study of SM backgrounds in  $Z \rightarrow \ell \ell$  and  $W \rightarrow \ell \nu$  analyses are also generated using PYTHIA6. The only exceptions are the  $t\bar{t}$  background and the  $W \rightarrow e\nu$  samples used in Section 8.2, which are generated with the MC@NLO program [15]. For the study of the total transverse energy of the events, samples produced with PYTHIA8 [16] are used as well.

MC samples were produced with different levels of pile-up in order to reflect the conditions in different data-taking periods. In particular, two event samples were used for jets: one was simulated with a pile-up model where only pile-up collisions originating from the primary bunch crossing are considered (in-time pile-up) and a second one was simulated with a realistic configuration of the LHC bunch group structure, where pile-up collisions from successive bunch crossings are also included in the simulation. In the case of events containing  $Z \rightarrow \ell \ell$  or  $W \rightarrow \ell \nu$ , MC samples with in-time pile-up configuration are used, because these data correspond to periods where the contribution of out-of-time pileup is small.

The trigger and event selection criteria used for the data are also applied to the MC simulation.

#### 5 $\boldsymbol{\textit{E}}_{T}^{miss}$ reconstruction and calibration

The  $E_{\rm T}^{\rm miss}$  reconstruction includes contributions from energy deposits in the calorimeters and muons reconstructed in the muon spectrometer. The two  $E_{\rm T}^{\rm miss}$  components are calculated as:

$$E_{x(y)}^{\text{miss}} = E_{x(y)}^{\text{miss,calo}} + E_{x(y)}^{\text{miss},\mu}.$$
 (1)

Low- $p_{\rm T}$  tracks are used to recover low  $p_{\rm T}$  particles which are missed in the calorimeters (see Section 5.3.1), and muons reconstructed from the inner detector are used to recover muons in regions not covered by the muon spectrometer (see Section 5.2). The two terms in the above equation are referred to as the calorimeter and muon terms, and will be described in more detail in the following sections. The values of  $E_{\rm T}^{\rm miss}$  and its azimuthal coordinate ( $\phi^{\rm miss}$ ) are then calculated as:

$$E_{\rm T}^{\rm miss} = \sqrt{\left(E_x^{\rm miss}\right)^2 + \left(E_y^{\rm miss}\right)^2} ,$$
  
$$\phi^{\rm miss} = \arctan(E_y^{\rm miss}, E_x^{\rm miss}). \tag{2}$$

#### 5.1 Calculation of the $\boldsymbol{E}_{T}^{miss}$ calorimeter term

In this paper, the  $E_{\rm T}^{\rm miss}$  reconstruction uses calorimeter cells calibrated according to the reconstructed physics object to which they are associated. Calorimeter cells are associated with a reconstructed and identified high- $p_{\rm T}$  parent object in a chosen order: electrons, photons, hadronically decaying  $\tau$ -leptons, jets

and muons. Cells not associated with any such objects are also taken into account in the  $E_{\rm T}^{\rm miss}$  calculation. Their contribution, named  $E_T^{\text{miss,CellOut}}$  hereafter, is important for the  $E_T^{\text{miss}}$  resolution [17].

Once the cells are associated with objects as described above, the  $E_{\rm T}^{\rm miss}$  calorimeter term is calculated as follows (note that the  $E_{x(y)}^{\mathrm{miss,calo},\mu}$  term is not always added, as explained in Section 5.2, and for that reason it is written between parentheses):

$$E_{x(y)}^{\text{miss,calo}} = E_{x(y)}^{\text{miss},e} + E_{x(y)}^{\text{miss},\gamma} + E_{x(y)}^{\text{miss},\tau} + E_{x(y)}^{\text{miss,jets}} + E_{x(y)}^{\text{miss,calo},\mu} + E_{x(y)}^{\text{miss,calo},\mu}) + E_{x(y)}^{\text{miss,CellOut}}$$
(3)

where each term is calculated from the negative sum of calibrated cell energies inside the corresponding objects, as:

$$E_x^{\text{miss,term}} = -\sum_{i=1}^{N_{\text{cell}}^{\text{term}}} E_i \sin \theta_i \cos \phi_i \quad ,$$
$$E_y^{\text{miss,term}} = -\sum_{i=1}^{N_{\text{cell}}^{\text{term}}} E_i \sin \theta_i \sin \phi_i \tag{4}$$

where  $E_i$ ,  $\theta_i$  and  $\phi_i$  are the energy, the polar angle and the azimuthal angle, respectively. The summations are over all cells associated with specified objects in the pseudorapidity range<sup>4</sup>  $|\eta| < 4.5.$ 

Because of the high granularity of the calorimeter, it is crucial to suppress noise contributions and to limit the cells used in the  $\hat{E}_{\mathrm{T}}^{\mathrm{miss}}$  sum to those containing a significant signal. This is achieved by using only cells belonging to threedimensional topological clusters, referred as topoclusters hereafter [18], with the exception of electrons and photons for which a different clustering algorithm is used [11]. The topoclusters are seeded by cells with deposited energy  $|E_i| > 4\sigma_{\text{noise}}$ , and are built by iteratively adding neighbouring cells with  $|E_i| >$  $2\sigma_{noise}$  and, finally, by adding all neighbours of the accumulated cells.

The various terms in Equation 3 are described in the following

- $E_{x(y)}^{\text{miss},e}$ ,  $E_{x(y)}^{\text{miss},\gamma}$ ,  $E_{x(y)}^{\text{miss},\tau}$  are reconstructed from cells in clusters associated to electrons, photons and  $\tau$ -jets from hadronically decaying  $\tau$ -leptons, respectively;
- $E_{x(y)}^{\text{miss,jets}}$  is reconstructed from cells in clusters associated to jets with calibrated  $p_{\rm T} > 20$  GeV;
- $E_{x(y)}^{\text{miss,softjets}}$  is reconstructed from cells in clusters associated to jets with 7 GeV  $< p_{\rm T} < 20$  GeV;
- $E_{x(y)}^{\text{miss,calo},\mu}$  is the contribution to  $E_{\text{T}}^{\text{miss}}$  originating from the energy lost by muons in the calorimeter (see Section 5.2); • the  $E_{x(y)}^{\text{miss,CellOut}}$  term is calculated from the cells in topoclus-
- ters which are not included in the reconstructed objects.

All these terms are calibrated independently as described in Section 5.3. The final  $E_{x(y)}^{\text{miss}}$  is calculated from Equation 1 adding the  $E_{x(y)}^{\text{miss},\mu}$  term, described in Section 5.2.

#### 5.2 Calculation of the $E_T^{miss}$ muon term

The  $E_{\rm T}^{\rm miss}$  muon term is calculated from the momenta of muon tracks reconstructed with  $|\eta| < 2.7$ :

$$E_{x(y)}^{\mathrm{miss},\mu} = -\sum_{\mathrm{muons}} p_{x(y)}^{\mu}$$
(5)

where the summation is over selected muons. In the region  $|\eta| < 2.5$ , only well-reconstructed muons in the muon spectrometer with a matched track in the inner detector are considered (combined muons). The matching requirement considerably reduces contributions from fake muons (reconstructed muons not corresponding to true muons). These fake muons can sometimes be created from high hit multiplicities in the muon spectrometer in events where some particles from very energetic jets punch through the calorimeter into the muon system.

In order to deal appropriately with the energy deposited by the muon in the calorimeters,  $E_{x(y)}^{\text{miss,calo},\mu}$ , the muon term is calculated differently for isolated and non-isolated muons, with non-isolated muons defined as those within a distance  $\Delta R =$  $\sqrt{(\Delta \eta)^2 + (\Delta \phi)^2} < 0.3$  of a reconstructed jet in the event:

- The  $p_{\rm T}$  of an isolated muon is determined from the combined measurement of the inner detector and muon spectrometer, taking into account the energy deposited in the calorimeters. In this case the energy lost by the muon in the calorimeters  $(E_{x(y)}^{\text{miss,calo},\mu})$  is not added to the calorimeter term (Equation 3) to avoid double counting of energy.
- For a non-isolated muon, the energy deposited in the calorimeter cannot be resolved from the calorimetric energy depositions of the particles in the jet. The muon spectrometer measurement of the muon momentum after energy loss in the calorimeter is therefore used, so the  $E_{x(y)}^{\text{miss,calo},\mu}$  term is added to the calorimeter term (Equation 3). Only in cases in which there is a significant mis-match between the spectrometer and the combined measurement, the combined measurement is used and a parameterized estimation of the muon energy loss in the calorimeter [10] is subtracted.

For higher values of pseudorapidity (2.5  $< |\eta| < 2.7$ ), outside the fiducial volume of the inner detector, there is no matched track requirement and the muon spectrometer  $p_{\rm T}$  alone is used for both isolated and non-isolated muons.

Aside from the loss of muons outside the acceptance of the muon spectrometer ( $|\eta| > 2.7$ ), muons can be lost in other small inactive regions (around  $|\eta| = 0$  and  $|\eta| \sim 1.2$ ) of the muon spectrometer. The muons which are reconstructed by segments matched to inner detector tracks extrapolated to the muon spectrometer are used to recover their contributions to  $E_{T}^{miss}$  in the  $|\eta| \sim 1.2$  regions [10].

Although the core of the  $E_{\rm T}^{\rm miss}$  resolution is not much affected by the muon term, any muons which are not reconstructed, badly measured, or fake, can be a source of fake  $E_{T}^{\text{miss}}$ .

#### 5.3 Calibration of $E_{T}^{miss}$

The calibration of  $E_{\rm T}^{\rm miss}$  is performed using the scheme described below, where the cells are calibrated separately according to their parent object:

This  $\eta$  cut is chosen because the MC simulation does not describe data well in the very forward region.

 $<sup>\</sup>sigma_{noise}$  is the Gaussian width of the EM cell energy distribution measured in randomly triggered events far from collision bunches.

- The  $E_{\rm T}^{{\rm miss},e}$  term is calculated from reconstructed electrons passing the "medium" electron identification requirements, with  $p_{\rm T} > 10$  GeV and calibrated with the default electron calibration [8].
- The  $E_{\rm T}^{\rm miss,\gamma}$  term is calculated from photons reconstructed with the "tight" photon identification requirements [11], with  $p_{\rm T} > 10$  GeV at the EM scale. Due to the low photon purity, the default photon calibration is not applied.
- The  $E_{\rm T}^{\rm miss,\tau}$  term is calculated from  $\tau$ -jets reconstructed with the "tight"  $\tau$ -identification requirements [19], with  $p_{\rm T} > 10$ GeV, calibrated with the local hadronic calibration (LCW) scheme [20]. The LCW scheme uses properties of clusters to calibrate them individually. It first classifies calorimeter clusters as electromagnetic or hadronic, according to the cluster topology, and then weights each calorimeter cell in clusters according to the cluster energy and the cell energy density. Additional corrections are applied to the cluster energy for the average energy deposited in the non-active material before and between the calorimeters and for unclustered calorimeter energy.
- The  $E_{\rm T}^{\rm miss, softjets}$  term is calculated from jets (reconstructed using the anti- $k_t$  algorithm with R=0.6) with 7 <  $p_{\rm T}$  < 20 GeV calibrated with the LCW calibration.
- The  $E_{\rm T}^{\rm miss,jets}$  term is calculated from jets with  $p_{\rm T} > 20$  GeV calibrated with the LCW calibration and the jet energy scale (JES) factor [21] applied. The JES factor corrects the energy of jets, either at the EM-scale or after cluster calibration, back to particle level. The JES is derived as a function of reconstructed jet  $\eta$  and  $p_{\rm T}$  using the generator-level information in MC simulation.
- The  $E_{\rm T}^{\rm miss,CellOut}$  term is calculated from topoclusters outside reconstructed objects with the LCW calibration and from reconstructed tracks as described in Section 5.3.1.

Note that object classification criteria and calibration can be chosen according to specific analysis criteria, if needed.

# 5.3.1 Calculation of the $E_{\rm T}^{\rm miss, CellOut}$ term with a track-cluster matching algorithm

In events with W and Z boson production, the calibration of the  $E_{\rm T}^{\rm miss, CellOut}$  term is of particular importance because, due to the low particle multiplicity in these events, this  $E_{\rm T}^{\rm miss}$  contribution balances the W/Z boson  $p_{\rm T}$  to a large extent [17]. An energy-flow algorithm is used to improve the calculation of the low- $p_{\rm T}$  contribution to  $E_{\rm T}^{\rm miss}$  ( $E_{\rm T}^{\rm miss, CellOut}$ ). Tracks are added to recover the contribution from low- $p_{\rm T}$  particles which do not reach the calorimeter or do not seed a topocluster. Furthermore the track momentum is used instead of the topocluster energy for tracks associated to topoclusters, thus exploiting the better calibration and resolution of tracks at low momentum compared to topoclusters.

Reconstructed tracks with  $p_{\rm T} > 400$  MeV, passing track quality selection criteria such as the number of hits and  $\chi^2$  of the track fit, are used for the calculation of the  $E_{\rm T}^{\rm miss,CellOut}$  term. All selected tracks are extrapolated to the second layer of the electromagnetic calorimeter and very loose criteria are used for association to reconstructed objects or topoclusters, to avoid double counting. If a track is neither associated to a topocluster nor a reconstructed object, its transverse momentum is added to the calculation of  $E_{\rm T}^{\rm miss,CellOut}$ . In the case where the track is associated to a topocluster, its transverse momentum is used for the calculation of the  $E_{\rm T}^{\rm miss,CellOut}$  and the topocluster energy is discarded, assuming that the topocluster energy corresponds to the charged particle giving the track. It has to be noticed that there is a strong correlation between the number of particles and topoclusters, so, in general no neutral energy is lost replacing the topocluster by a track, and the neutral topoclusters are kept in most of the cases. If more than one topocluster is associated to a track, only the topocluster with the largest energy is excluded from the  $E_{\rm T}^{\rm miss}$  calculation, assuming that this energy corresponds to the track.

#### 6 Study of $E_{\rm T}^{\rm miss}$ performance

In this section the distributions of  $E_{\rm T}^{\rm miss}$  in minimum bias, dijet,  $Z \to \ell \ell$  and  $W \to \ell \nu$  events from data are compared with the expected distributions from the MC samples. The performance of  $E_{\rm T}^{\rm miss}$  in terms of resolution and scale is also derived.

Minimum bias, di-jet events and  $Z \rightarrow \ell \ell$  events are used to investigate the  $E_{\rm T}^{\rm miss}$  performance without relying on MC detector simulation. In general, apart from a small contribution from the semi-leptonic decay of heavy-flavour hadrons in jets, no genuine  $E_{\rm T}^{\rm miss}$  is expected in these events. Thus most of the  $E_{\rm T}^{\rm miss}$  reconstructed in these events is a direct result of imperfections in the reconstruction process or in the detector response.

### 6.1 $\boldsymbol{E}_{T}^{miss}$ performance in minimum bias and di-jet events

The distributions of  $E_x^{\text{miss}}$ ,  $E_y^{\text{miss}}$ ,  $E_T^{\text{miss}}$  and  $\phi^{\text{miss}}$  for data and MC simulation are shown in Figure 1 for minimum bias events. The distributions are shown only for events with total transverse energy (see definition at the end of this section) greater than 20 GeV in order to reduce the contamination of fake triggers from the MBTS. Figure 2 shows the distributions of the same variables for the di-jet sample. The di-jet sample corresponding to the periods with higher pileup conditions (see Section 3.1) is used. The MC simulation expectations are superimposed, normalized to the number of events in the data.

In di-jet events a reasonable agreement is found between data and simulation for all basic quantities, while there is some disagreement in minimum bias events, attributed to imperfect modelling of soft particle activity in the MC simulation. The better agreement between data and MC simulation in the  $\phi^{\text{miss}}$  distribution for the di-jet sample can be partly explained by the fact that the  $E_{\text{T}}^{\text{miss}}$  is not corrected for the primary vertex position; the primary vertex position in data is better reproduced by the MC simulation for the di-jet sample than in the case of the minimum bias sample.

Events in the tails of the  $E_T^{\text{miss}}$  distributions have been carefully checked, in order to understand the origin of the large measured  $E_T^{\text{miss}}$ . The tails are not completely well described by MC simulation, but, both in data and in MC simulation they are

in general due to mis-measured jets. In minimum bias events there are more events in the tail in MC simulation and this can be due to the fact that the MC statistics is larger than in data. In di-jet events, there are more events in the tail in data. More MC events would be desirable. In di-jet events there are 19 events with  $E_T^{\text{miss}} > 110$  GeV in the data. The majority of them (13 events) are due to mis-measured jets, where in most of the cases at least one jet points to a transition region between calorimeters. Two events are due to a combination of mis-measured jets with an overlapping muon, and one event is due to a fake high $p_T$  muon. Finally two events look like good  $b\bar{b}$  candidates, and one event has one reconstructed jet and no activity in the other hemisphere.

The events with fake  $E_{\rm T}^{\rm miss}$  due to mis-measured jets and jets containing leptonic decays of heavy hadrons can be rejected by a cut based on the azimuthal angle between the jet and  $E_{\rm T}^{\rm miss}$ ,  $\Delta \phi$ (jet, $E_{\rm T}^{\rm miss}$ ). Since the requirement of event cleaning depends on the physics analysis, the minimal cleaning cut is applied and careful evaluation of tail events is performed in this paper. Analyses that rely on a careful understanding and reduction of the tails of the  $E_{\rm T}^{\rm miss}$  distribution (e.g. SUSY searches such as Ref. [3]) have performed more detailed studies to characterize the residual tail in events containing high- $p_{\rm T}$  jets. These analyses use tighter jet cleaning cuts, track-jet matching, and angular cuts on  $\Delta \phi$ (jet, $E_{\rm T}^{\rm miss}$ ) to further reduce the fake  $E_{\rm T}^{\rm miss}$  tail. In Ref. [3] a fully data-driven method (described in detail in Ref. [17]) was then employed to determine the residual fake  $E_{\rm T}^{\rm miss}$  background.

The contributions from jets, soft jets and topoclusters not associated to the reconstructed objects and muons are shown in Figure 3 for the di-jet events. The data-MC agreement is good for all of the terms contributing to  $E_{\rm T}^{\rm miss}$ . The tails observed in the muon term are mainly due to reconstructed fake muons and to one cosmic-ray muon, which can be rejected by applying a tighter selection for the muons used in the  $E_{\rm T}^{\rm miss}$  reconstruction, based on  $\chi^2$  criteria for the combination, isolation criteria and requirements on the number of hits in muon chambers used for the muon reconstruction.

In the following some distributions are shown for the total transverse energy,  $\sum E_{\rm T}$ , which is an important quantity to parameterise and understand the  $E_{\rm T}^{\rm miss}$  performance. It is defined as:

$$\sum E_{\rm T} = \sum_{i=1}^{N_{\rm cell}} E_i \sin \theta_i \tag{6}$$

where  $E_i$  and  $\theta_i$  are the energy and the polar angle, respectively, of calorimeter cells associated to topoclusters within  $|\eta| < 4.5$ . Cell energies are calibrated according to the scheme described in Section 5.3 for  $E_{\rm T}^{\rm miss}$ .

The data distributions of  $\sum E_T$  for minimum bias and di-jet events from the subset corresponding to lower pileup conditions (see Section 3.1) are compared to MC predictions from two versions of PYTHIA in Figure 4. The left-hand distributions show comparisons with the ATLAS tune of PYTHIA6. The right-hand distributions show the comparisons with the default tune of PYTHIA8. Due to the limited number of events simulated, the distribution for the di-jet PYTHIA8 MC sample is not smooth, and is zero in the lowest  $\sum E_T$  bin populated by data. This is not understood, also if it can be partly explained by the fact that the low  $\sum E_{\rm T}$  region is populated by events from the jet MC sample generated in the lowest parton  $p_{\rm T}$  bin (17-35 GeV), which is the most suppressed by the di-jet selection (a factor about 20 more than other samples) and has a large weight, due to cross-section. Moreover the PYTHIA8 jet MC sample in the 8-17 GeV parton  $p_{\rm T}$  bin is not available. In the case of the minimum bias sample, due to the very limited number of events simulated (about a factor 25 less respect to data), the tails in the PYTHIA8 MC distribution are strongly depleted.

The PYTHIA8 MC [16] version used in this paper has not yet been tuned to the ATLAS data. The current tune [22] uses the CTEQ 6.1 parton distribution functions (PDF) instead of the MRST LO<sup>\*\*</sup> as used in PYTHIA6, and its diffraction model differs, including higher- $Q^2$  diffractive processes. The comparison of the mean values and the shapes of the two different MC distributions with data seems to indicate that a better agreement is obtained with the PYTHIA8 but, due to the reduced PYTHIA8 MC statistics, no firm conclusion can be drawn. In the rest of the paper, the PYTHIA6 MC samples with the ATLAS tune are used for comparison with data; this version is used as the baseline for PYTHIA MC samples for 2010 data analyses.

#### 6.2 $\boldsymbol{E}_{T}^{miss}$ performance in $\boldsymbol{Z} \rightarrow \ell \ell$ events

The absence of genuine  $E_{\rm T}^{\rm miss}$  in  $Z \to \ell \ell$  events, coupled with the clean event signature and the relatively large cross-section, means that it is a good channel to study  $E_{\rm T}^{\rm miss}$  performance.

The distributions of  $E_{\rm T}^{\rm miss}$  and  $\phi^{\rm miss}$  for data and MC simulation are shown in Figure 5 for  $Z \rightarrow ee$  and  $Z \rightarrow \mu\mu$  events. The contributions due to muons are shown for  $Z \rightarrow \mu \mu$  events in Figure 6. Both the contributions from energy deposited in calorimeter cells associated to muons, taken at the EM scale, and the contributions from reconstructed muons are shown. For  $Z \rightarrow ee$  events, the contributions from electrons, jets, soft jets and topoclusters outside the reconstructed objects are shown separately in Figure 7. The peak at zero in the distribution of the jet term corresponds to events where there are no jets with  $p_{\rm T}$  above 20 GeV, and the small values (< 20 GeV) in the distribution are due to events with two jets whose transverse momenta balance. The MC simulation expectations, from  $Z \rightarrow \ell \ell$  events and from the dominant SM backgrounds, are superimposed. Each MC sample is weighted with its corresponding cross-section and then the total MC expectation is normalized to the number of events in data. Reasonable agreement between data and MC simulation is observed in all distributions.

Events in the tails of the  $E_{\rm T}^{\rm miss}$  distributions in Figure 5 have been carefully checked. The 22 events with the highest  $E_{\rm T}^{\rm miss}$ values, above 60 GeV, have been examined in detail to check whether they are related to cosmic-ray muon background, fake muons, badly measured jets or jets pointing to dead calorimeter regions. The events in the tails are found to be compatible with either signal candidates, including  $t\bar{t}$ , WW and WZ diboson events, all involving real  $E_{\rm T}^{\rm miss}$ , or events in which the  $E_{\rm T}^{\rm miss}$  vector is close to a jet in the transverse plane. The latter category of events can arise from mis-measured jets, and be rejected at the analysis level with cuts on  $\Delta \phi$ (**jet**,  $E_{\rm T}^{\rm miss}$ ) (see Section 6.1).



**Fig. 1.** Distribution of  $E_x^{\text{miss}}$  (top left),  $E_y^{\text{miss}}$  (top right),  $E_T^{\text{miss}}$  (bottom left),  $\phi^{\text{miss}}$  (bottom right) as measured in a data sample of minimum bias events. The expectation from MC simulation, normalized to the number of events in data, is superimposed.

#### 6.2.1 Measuring $\boldsymbol{E}_{T}^{miss}$ response in $Z \rightarrow \ell \ell$ events

From the event topology [17] in events with  $Z \rightarrow \ell \ell$  decay one can define an axis in the transverse plane such that the component of  $E_{\rm T}^{\rm miss}$  along this axis is sensitive to detector resolution and biases. The direction of this axis,  $A_{\rm Z}$ , is defined by the reconstructed momenta of the leptons:

$$\boldsymbol{A}_{\mathbf{Z}} = (\boldsymbol{p}_{\mathbf{T}}^{\ell^+} + \boldsymbol{p}_{\mathbf{T}}^{\ell^-}) / |\boldsymbol{p}_{\mathbf{T}}^{\ell^+} + \boldsymbol{p}_{\mathbf{T}}^{\ell^-}|$$
(7)

where  $p_{T}^{\ell}$  are the vector transverse momenta of the lepton and anti-lepton. The direction of  $A_{Z}$  thus reconstructs the direction of motion of the Z boson. The perpendicular axis in the transverse plane,  $A_{AZ}$ , is a unit vector placed at right angles to  $A_{Z}$ , with positive direction anticlockwise from the direction of the Z boson.

The mean value of the projection of  $E_{\rm T}^{\rm miss}$  onto the longitudinal axis,  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$ , is a measure of the  $E_{\rm T}^{\rm miss}$  scale, as this axis is sensitive to the balance between the leptons and the hadronic recoil. Figure 8 shows the value of  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$  as a function of  $p_{\rm T}^Z$ . These mean values are used as a diagnostic to validate the  $E_{\rm T}^{\rm miss}$  reconstruction algorithms. If the leptons perfectly balanced the hadronic recoil, regardless of the net momentum of the lepton system, then the  $E_{\rm T}^{\rm miss} \cdot A_{\rm Z}$  would be zero, independent of  $p_{\rm T}^Z$ . Instead,  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$  displays a small bias in both the electron and muon channels which is reasonably reproduced by the MC simulation. The observed bias is slightly negative for low values of  $p_T^Z$ , suggesting either that the  $p_T$  of the lepton system is overestimated or that the magnitude of the hadronic recoil is underestimated. The same sign and magnitude of bias is seen in both electron and muon channels, suggesting that the hadronic recoil, here dominated by  $E_T^{\text{miss,CellOut}}$  and by soft jets, is the source of bias. The component of the  $E_T^{\text{miss}}$  along the perpendicular axis,  $E_T^{\text{miss}} \cdot A_{\text{AZ}}$ , displays no bias, and, indeed there is no mechanism for generating such a bias.

In Figure 9 the dependences of  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$  on  $p_{\rm T}^{Z}$  are shown separately for events with  $Z \rightarrow \ell \ell$  produced in association with zero jets or with at least one jet, with the jet definition as described in Section 3.1. The figure demonstrates that there is a negative bias in  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$  for events with zero jets, which increases with  $p_{\rm T}^{Z}$  up to 6 GeV. A similar bias is observed in both electron and muon channels, hence it is interpreted as coming from imperfections in the calibration of the soft hadronic recoil (the  $E_{\rm T}^{\rm miss,CellOut}$  and the  $E_{\rm T}^{\rm miss,softjets}$  terms). In events with at least one jet there is a small positive bias in the electron channel at high  $p_{\rm T}^{Z}$ , which is visible also in the muon channel for  $p_{\rm T}^{Z}$  in the region 15-20 GeV.



**Fig. 2.** Distribution of  $E_x^{\text{miss}}$  (top left),  $E_y^{\text{miss}}$  (top right),  $E_T^{\text{miss}}$  (bottom left),  $\phi^{\text{miss}}$  (bottom right) as measured in the data sample of di-jet events. The expectation from MC simulation, normalized to the number of events in data, is superimposed. The events in the tails are discussed in the text.

Figure 10 shows  $\langle E_{\rm T}^{\rm miss} \cdot A_{\rm Z} \rangle$  for  $Z \to \ell \ell$  events where there are neither high  $p_{\rm T}$  nor soft jets, for two cases of  $E_{\rm T}^{\rm miss}$  reconstruction: calculating the  $E_{\rm T}^{\rm miss,CellOut}$  term with the track-cluster matching algorithm (see Section 5.3.1) or calculating this term from the calorimeter topoclusters only (denoted as  $E_{\rm T}^{\rm miss}$  no tracks). The plots show a lower bias for the case with the track-cluster matching algorithm, indicating that it improves the reconstruction of the  $E_{\rm T}^{\rm miss,CellOut}$  term.

#### 6.3 $\boldsymbol{E}_{T}^{miss}$ performance in $\boldsymbol{W} \rightarrow \ell \boldsymbol{v}$ events

In this section the  $E_{\rm T}^{\rm miss}$  performance is studied in  $W \to ev$  and  $W \to \mu v$  events. In these events genuine  $E_{\rm T}^{\rm miss}$  is expected due to the presence of the neutrino, therefore the  $E_{\rm T}^{\rm miss}$  scale can be checked.

The distributions of  $E_{\rm T}^{\rm miss}$  and  $\phi^{\rm miss}$  in data and in MC simulation are shown in Figure 11 for  $W \to ev$  and  $W \to \mu v$  events. The contributions due to muons are shown for  $W \to \mu v$  events in Figure 12. Both, the  $E_{\rm T}^{\rm miss}$  contribution from energy deposited in calorimeter cells associated to muons, taken at the EM scale, and the  $E_{\rm T}^{\rm miss}$  contribution from reconstructed muons are shown. The contributions given by the electrons, jets, soft jets and topoclusters outside reconstructed

objects are shown in Figure 13 for  $W \rightarrow ev$  events. The MC expectations are also shown, both from  $W \rightarrow \ell v$  events, and from the dominant SM backgrounds. The MC simulation describes all of the quantities well, with the exception that very small data-MC discrepancies are observed in the distribution of the  $E_{\rm T}^{\rm miss,e}$  at low  $E_{\rm T}^{\rm miss}$  values. This can be attributed to the QCD jet background, which would predominantly populate the region of low  $E_{\rm T}^{\rm miss}$  [8], but which is not included in the MC expectation shown.

#### 6.3.1 $\boldsymbol{E}_{T}^{miss}$ linearity in $W \rightarrow \ell \nu$ MC events

The expected  $E_{\rm T}^{\rm miss}$  linearity, which is defined as the mean value of the ratio:  $(E_{\rm T}^{\rm miss} - E_{\rm T}^{\rm miss,{\rm True}})/E_{\rm T}^{\rm miss,{\rm True}}$ , is shown as a function of  $E_{\rm T}^{\rm miss,{\rm True}}$  in Figure 14 for  $W \to ev$  and  $W \to \mu v$  MC events. The mean value of this ratio is expected to be zero if the reconstructed  $E_{\rm T}^{\rm miss}$  has the correct scale. In Figure 14, it can be seen that there is a displacement from zero which varies with the true  $E_{\rm T}^{\rm miss}$ . The bias at low  $E_{\rm T}^{\rm miss,{\rm True}}$  values is about 5% and is due to the finite resolution of the  $E_{\rm T}^{\rm miss}$  measurement. The reconstructed  $E_{\rm T}^{\rm miss}$  is positive by definition, so the relative difference is positive when the  $E_{\rm T}^{\rm miss,{\rm True}}$  is small.



**Fig. 3.** Distribution of  $E_T^{\text{miss}}$  computed with cells from topoclusters in jets (top left), in soft jets (top right), from topoclusters outside reconstructed objects (bottom left) and from reconstructed muons (bottom right) for data for di-jet events. The expectation from MC simulation, normalized to the number of events in data, is superimposed. The events in the tail of the  $E_T^{\text{miss},\mu}$  distribution are discussed in the text.

The effect extends up to 40 GeV. The bias is in general larger for  $W \to \mu \nu$  events than for  $W \to e\nu$  events. Considering only events with  $E_{\rm T}^{\rm miss, True} > 40$  GeV, the  $E_{\rm T}^{\rm miss}$  linearity is better than 1% in  $W \to e\nu$  events, while there is a non-linearity up to about 3% in  $W \to \mu \nu$  events. This may be explained by an underestimation of the  $E_{\rm T}^{\rm miss, calo, \mu}$  term, in which too few calorimeter cells are associated to the reconstructed muon.

#### 6.4 $E_{\rm T}^{\rm miss}$ resolution

A more quantitative evaluation of the  $E_T^{\text{miss}}$  performance can be obtained from a study of the  $(E_x^{\text{miss}}, E_y^{\text{miss}})$  resolutions as a function of  $\sum E_T$ . In  $Z \to \ell \ell$  events, as well as in minimum bias and QCD jet events, no genuine  $E_T^{\text{miss}}$  is expected, so the resolution of the two  $E_T^{\text{miss}}$  components is measured directly from reconstructed quantities, assuming that the true values of  $E_x^{\text{miss}}$  and  $E_y^{\text{miss}}$  are equal to zero. The resolution is estimated from the width of the combined distribution of  $E_x^{\text{miss}}$  and  $E_y^{\text{miss}}$ (denoted  $(E_x^{\text{miss}}, E_y^{\text{miss}})$  distribution) in bins of  $\sum E_T$ . The core of the distribution is fitted, for each  $\sum E_T$  bin, with a Gaussian over twice the expected resolution obtained from previous studies [17] and the fitted width,  $\sigma$ , is examined as a function of  $\sum E_{\rm T}$ . The  $E_{\rm T}^{\rm miss}$  resolution follows an approximately stochastic behaviour as a function of  $\sum E_{\rm T}$ , which can be described with the function  $\sigma = k \cdot \sqrt{\sum E_{\rm T}}$ , but deviations from this simple law are expected in the low  $\sum E_{\rm T}$  region due to noise and in the very large  $\sum E_{\rm T}$  region due to the constant term.

Figure 15 (left) shows the resolution from data at  $\sqrt{s} = 7$ TeV for  $Z \to \ell \ell$  events, minimum bias and di-jet events as a function of the total transverse energy in the event, obtained by summing the  $p_T$  of muons and the  $\sum E_T$  in calorimeters, calculated as described in Section 6.1. If the resolution is shown as a function of the  $\sum E_T$  in calorimeters, a difference between  $Z \to ee$  and  $Z \to \mu\mu$  events is observed due to the fact that  $\sum E_T$  includes electron momenta in  $Z \to ee$  events while muon momenta are not included in  $Z \to \mu\mu$  events.

The resolution of the two  $E_T^{\text{miss}}$  components is fitted with the simple function given above. The fits are acceptable and are of similar quality for all different channels studied. This allows to use the parameter *k* as an estimator for the resolution and to compare it in various physics channels in data and MC simulation. There is a reasonable agreement in the  $E_T^{\text{miss}}$  resolution in the different physics channels, as can be seen from the fit parameters *k* reported in the figure. The *k* parameter has fit values ranging from 0.42 GeV<sup>1/2</sup> for  $Z \rightarrow \ell \ell$  events to 0.51 GeV<sup>1/2</sup> for di-jet events. The  $E_T^{\text{miss}}$  resolution is better in  $Z \rightarrow \ell \ell$  events



**Fig. 4.** Distribution of  $\sum E_T$  as measured in a data sample of minimum bias events (top) and di-jet events (bottom) selecting two jets with  $p_T > 25$  GeV. The expectation from MC simulation, normalized to the number of events in data, is superimposed. On the left PYTHIA6 (ATLAS tune) is compared with the data. On the right PYTHIA8 is compared with the data.

because the lepton momenta are measured with better precision than jets.

In Figure 15 (right) the  $E_T^{\text{miss}}$  resolution is shown for MC events. In addition to the  $Z \to \ell \ell$ , minimum bias and di-jet events, the resolution is also shown for  $W \to \ell \nu$  MC events. In W events the resolution of the two  $E_T^{\text{miss}}$  components is estimated from the width of  $(E_x^{\text{miss}} - E_x^{\text{miss,True}}, E_y^{\text{miss}} - E_y^{\text{miss,True}})$ in bins of  $\Sigma E_T$ , fitted with a Gaussian as explained above. There is a reasonable agreement in the  $E_T^{\text{miss}}$  resolution in the different MC channels studied with the fitted value of k ranging from 0.42 GeV<sup>1/2</sup> for  $Z \to \ell \ell$  events to 0.50 GeV<sup>1/2</sup> for di-jet events. As observed for data, the  $E_T^{\text{miss}}$  resolution is better in  $Z \to \ell \ell$  events and slightly better in  $W \to \ell \nu$  events, due to the presence of the leptons which are more precisely measured.

The resolution in MC minimum bias events is slightly worse than in data. This is probably due to imperfections of the modelling of soft particle activity in MC simulation, while there is a good data-MC agreement in the resolution for other channels.

# 7 Evaluation of the systematic uncertainty on the $\boldsymbol{E}_{T}^{miss}$ scale

For any analysis using  $E_{\rm T}^{\rm miss}$ , it is necessary to be able to evaluate the systematic uncertainty on the  $E_{\rm T}^{\rm miss}$  scale. The  $E_{\rm T}^{\rm miss}$ , as defined in Section 5.3, is the sum of several terms corresponding to different types of reconstructed objects. The uncertainty on each individual term can be evaluated given the knowledge of the reconstructed objects [8,23] that are used to build it and this uncertainty can be propagated to  $E_{\rm T}^{\rm miss}$ . The overall systematic uncertainty on the  $E_{\rm T}^{\rm miss}$  scale is then calculated by combining the uncertainties on each term.

The relative impact of the uncertainty of the constituent terms on  $E_{\rm T}^{\rm miss}$  differs from one analysis to another depending on the final state being studied. In particular, in events containing *W* and *Z* bosons decaying to leptons, uncertainties on the scale and resolution in the measurements of the charged leptons, together with uncertainties on the jet energy scale, need to be propagated to the systematic uncertainty estimate of  $E_{\rm T}^{\rm miss}$ . Another significant contribution to the  $E_{\rm T}^{\rm miss}$  scale uncertainty in *W* and *Z* boson final states comes from the contribution of topoclusters outside reconstructed objects and from soft jets. In the next three subsections, two complementary methods for the evaluation of the systematic uncertainty on the  $E_{\rm T}^{\rm miss, CellOut}$  and



**Fig. 5.** Distribution of  $E_T^{\text{miss}}$  (top) and  $\phi^{\text{miss}}$ (bottom) as measured in a data sample of  $Z \rightarrow ee$  (left) and of  $Z \rightarrow \mu\mu$  (right). The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section. The sum of all backgrounds is shown in the lower plots.



**Fig. 6.** Distribution of  $E_T^{\text{miss}}$  computed with calorimeter cells associated to muons  $(E_T^{\text{miss,calo},\mu})$  (left) and computed from reconstructed muons  $(E_T^{\text{miss},\mu})$  (right) for  $Z \to \mu\mu$  data. The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section.



**Fig. 7.** Distribution of  $E_T^{\text{miss}}$  computed with cells associated to electrons  $(E_T^{\text{miss},e})$  (top left), jets with  $p_T > 20 \text{ GeV}(E_T^{\text{miss,jets}})$  (top right), jets with 7 GeV  $< p_T < 20 \text{ GeV}(E_T^{\text{miss,softjets}})$  (bottom left) and from topoclusters outside reconstructed objects  $(E_T^{\text{miss,CellOut}})$  (bottom right) for  $Z \rightarrow ee$  data. The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section.

the  $E_{\rm T}^{\rm miss, softjets}$  terms are described. Finally the overall  $E_{\rm T}^{\rm miss}$  uncertainty for  $W \to \ell v$  events is calculated.  $E_{\rm T}^{\rm miss, CellOut}$  term, as verified in Ref. [24]. In order to cross-check for a possible dependence on the event total transverse

### 7.1 Evaluation of the systematic uncertainty on the $E_T^{miss,CellOut}$ scale using Monte Carlo simulation

There are several possible sources of systematic uncertainty in the calculation of  $E_{\rm T}^{\rm miss, CellOut}$ . These sources include inaccuracies in the description of the detector material, the choice of shower model and the model for the underlying event in the simulation. The systematic uncertainty due to each of these sources is estimated with dedicated MC simulations. The MC jet samples, generated with PYTHIA, are those used to assess the systematic uncertainty on the jet energy scale [21]. Table 1 lists the simulation samples considered, referred to in the following as "variations" with respect to the nominal sample. The estimate of the uncertainty on  $E_{\rm T}^{\rm miss, CellOut}$  for a variation.

The estimate of the uncertainty on  $E_{\rm T}^{\rm miss, centout}$  for a variation *i* is determined by calculating the percentage difference between the mean value of this term for the nominal sample, labelled  $\mu_0$ , and that for the variation sample, labelled  $\mu_i$ . This approach assumes that the variations affect the total scale and none of the variations introduces a shape dependence in the  $E_{\rm T}^{\rm mss, CellOut}$  term, as verified in Ref. [24]. In order to crosscheck for a possible dependence on the event total transverse energy, the relative difference  $R_i = (\mu_i - \mu_0)/\mu_0$  between different variations is computed in bins of  $\sum E_{\rm T}$  for the jet samples. No significant dependence of  $R_i$  on  $\sum E_{\rm T}$  is observed. A crosscheck on the topology dependence is done using  $W \rightarrow \ell v$  samples simulated by introducing the variations *i*. Table 2 shows the  $R_i$  values as computed in both the QCD jet samples and the  $W \rightarrow \ell v$  samples. The results are consistent, showing that the estimated uncertainty does not have a large dependence on the event topology.

A symmetric systematic uncertainty on the  $E_{\rm T}^{\rm miss,CellOut}$  scale is obtained by summing in quadrature the estimated uncertainties averaged between simulated jet and W events. The total estimated uncertainty<sup>6</sup> on the  $E_{\rm T}^{\rm miss,CellOut}$  term is 2.6%.

<sup>&</sup>lt;sup>6</sup> In this uncertainty evaluation using MC simulation, the uncertainty on the absolute electromagnetic energy scale in the calorimeters should also be taken into account. For the bulk of the LAr barrel electromagnetic calorimeter a 1.5% uncertainty is found on the cell energy measurement, increasing to 5% for the presampler and 3% for the tile calorimeter [25].



**Fig. 8.** Mean values of  $E_T^{\text{miss}} \cdot A_Z$  as a function of  $p_T^Z$  in  $Z \to ee$  (left) and  $Z \to \mu \mu$  (right) events.



**Fig. 9.** Mean value of  $E_T^{\text{miss}} \cdot A_Z$  as a function of  $p_T^Z$  requiring either zero jets with  $p_T > 20$  GeV or at least 1 jet with  $p_T > 20$  GeV in the event for  $Z \rightarrow ee$  (left) and  $Z \rightarrow \mu\mu$  (right) events.

Variation	Description
Dead Material	5% increase in the inner detector material
	0.1 $X_0$ in front of the cryostat of the EM barrel calorimeter
	0.05 $X_0$ between presampler and EM barrel calorimeter
	0.1 $X_0$ in the cryostat after the EM barrel calorimeter
	density of material in barrel-endcap transition of the EM calorimeter $\times 1.5$
FTFP_BERT	An alternative shower model for hadronic interaction in GEANT4
QGSP	An alternative shower model for hadronic interaction in GEANT4
PYTHIA Perugia 2010 tune	An alternative setting of the PYTHIA parameters
	with increased final state radiation and more soft particles

**Table 1.** Variations of the default simulation settings used for the estimate of the  $E_T^{\text{miss,CellOut}}$  term systematic uncertainty. See Ref. [21] for details of the parameters.

# 7.2 Evaluation of the systematic uncertainty on the $E_T^{miss,CellOut}$ scale from the topocluster energy scale uncertainty

The uncertainty on the scale of the  $E_{\rm T}^{\rm miss, CellOut}$  term, which is built from topoclusters with a correction based on tracks (see Section 5.3.1), can also be calculated from the topocluster energy scale uncertainties. These uncertainties can be estimated from comparisons between data and MC simulation using the E/p response from single tracks, measured by summing the energies of all calorimeter clusters around a single isolated track [25]. The effects of these uncertainties on the  $E_{\rm T}^{\rm miss,CellOut}$  term can be evaluated by varying the energy scale of topoclusters that contribute to the  $E_{\rm T}^{\rm miss,CellOut}$  term in  $W \rightarrow ev$  MC samples, as was done in Ref. [8].



**Fig. 10.** Mean value of  $E_T^{\text{miss}} \cdot A_Z$  as a function of  $p_T^Z$  in  $Z \to ee$  (left) and  $Z \to \mu \mu$  (right) for events with no jets with  $p_T > 7$  GeV. The default  $E_T^{\text{miss}}$  is compared with  $E_T^{\text{miss}}$  calculated in the same way with the exception that the track-cluster matching algorithm is not used for the calculation of  $E_T^{\text{miss,CellOut}}$ .



**Fig. 11.** Distribution of  $E_T^{\text{miss}}$  (top) and  $\phi^{\text{miss}}$  (bottom) as measured in a data sample of  $W \to ev$  (left) and  $W \to \mu v$  (right) events. The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section. The sum of all backgrounds is shown in the lower plots.



**Fig. 12.** Distribution of  $E_{\rm T}^{\rm miss}$  computed with cells from muons  $(E_{\rm T}^{\rm miss, calo, \mu})$  (left) and reconstructed muons  $(E_{\rm T}^{\rm miss, muon})$  (right) for  $W \rightarrow \mu v$  data. The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section.



**Fig. 13.** Distribution of  $E_T^{\text{miss}}$  computed with cells associated to electrons  $(E_T^{\text{miss},e})$  (top left), jets with  $p_T > 20 \text{ GeV}(E_T^{\text{miss,jets}})$  (top right), jets with  $p_T < 20 \text{ GeV}(E_T^{\text{miss,softjets}})$  (bottom left) and from topoclusters outside reconstructed objects  $(E_T^{\text{miss,CellOut}})$  (bottom right) for data. The expectation from Monte Carlo simulation is superimposed and normalized to data, after each MC sample is weighted with its corresponding cross-section.



**Fig. 14.**  $E_{\rm T}^{\rm miss}$  linearity in  $W \to ev$  and  $W \to \mu v$  MC events as a function of  $E_{\rm T}^{\rm miss, True}$ 



**Fig. 15.**  $E_x^{\text{miss}}$  and  $E_y^{\text{miss}}$  resolution as a function of the total transverse energy in the event calculated by summing the  $p_T$  of muons and the total transverse energy in the calorimeter in data at  $\sqrt{s} = 7 \text{ TeV}$  (left) and MC (right). The resolution of the two  $E_T^{\text{miss}}$  components is fitted with a function  $\sigma = k \cdot \sqrt{\Sigma E_T}$  and the fitted values of the parameter k, expressed in  $GeV^{1/2}$ , are reported in the figure.

Variation	jet events	W production	
Dead Material	$(-0.5\pm0.1)\%$	$(-0.6\pm0.2)\%$	
FTFP_BERT	$(0.1 \pm 0.4)\%$	$(0.5 \pm 0.2)\%$	
QGSP	$(-1.6 \pm 0.4)\%$	$(-2.2\pm0.2)\%$	
PYTHIA Perugia 2010 tune	$(-1.7 \pm 0.1)\%$	$(-1.5\pm0.2)\%$	

**Table 2.** Systematic uncertainties  $(R_i)$  on  $E_T^{\text{miss,CellOut}}$  associated with variations in the dead material (all the variations listed in Table 1 are applied at the same time), in the calorimeter shower modelling (FTFP\_BERT, QGSP) and in the event generator settings (PYTHIA Perugia 2010 tune).

The shift in the topocluster energy scale is applied by multiplying the topocluster energy by the function:

$$1 \pm a \times (1 + b/p_T) \tag{8}$$

with a = 3(10)% for  $|\eta| < (>)3.2$  and b = 1.2 GeV.

The *a* parameter in Equation 8 addresses the uncertainty on the cluster energy scale, obtained by comparing the ratio of the cluster energy and the measured track momentum, E/p, in data and MC simulation [25]. The value in the forward region, where tracks cannot be used to validate the energy scale, is estimated from the transverse momentum balance of one jet in the central region and one jet in the forward region in events with only two jets at high transverse momenta.

The *b* parameter in Equation 8 addresses the possible change in the clustering efficiency and scale in a non-isolated environment. To go from the response for single isolated particles to the cluster energy scale, possible effects from the noise thresholds in the configuration with nearby particles are taken into account .

Because of threshold effects, more energy is clustered for nearby particles than for isolated ones. In an hypothetic worst case scenario, the environment is so busy that the clustering algorithm is forced to cluster all the deposited energy, with no bias due to the noise thresholds. Therefore, the maximal size of the noise threshold effect can be evaluated by comparing the ratio  $E_{cell}/p$  of the total energy  $E_{cell}$  deposited into all cells around an isolated track to the track momentum, to the ratio E/p of the clustered energy E to the track momentum, in data and MC simulation. The fractional  $E_{\rm T}^{\rm miss, CellOut}$  uncertainty is evaluated from:

$$(\Delta^{\text{CellOut+}} + \Delta^{\text{CellOut-}}) / (2 \times E_{\text{T}}^{\text{miss,CellOut}})$$
(9)

where

$$\Delta^{\text{CellOut+}} = |E_{\text{T}}^{\text{miss,CellOut+}} - E_{\text{T}}^{\text{miss,CellOut}}|$$
$$\Delta^{\text{CellOut-}} = |E_{\text{T}}^{\text{miss,CellOut-}} - E_{\text{T}}^{\text{miss,CellOut}}|$$
(10)

with  $E_{\rm T}^{\rm miss, CellOut+}$  and  $E_{\rm T}^{\rm miss, CellOut-}$  obtained by shifting the topocluster energies up and down, respectively, using Equation 8. The value of the fractional  $E_T^{miss,CellOut}$  uncertainty is found to be approximately 13%, decreasing slightly with increasing  $\sum E_{\rm T}^{\rm CellOut}$ . This uncertainty is much larger than the uncertainty due to the detector description estimated from the first three lines of Table 2. The main reason is that the values of a and b which enter into Equation 8 are conservative, to include the effects described above. In particular the cluster energy uncertainty in the forward region is conservatively estimated, since the uncertainty cannot be evaluated using tracks. Moreover, the procedure does not take into account the fact that when the clusters are shifted up in  $p_{\rm T}$ , some of them can form jets above threshold and they are therefore included in the soft jet term in  $E_{\rm T}^{\rm miss}$ . These clusters should be removed from the  $E_{\rm T}^{\rm miss, CellOut}$ , they are in fact kept and this increases the uncertainty. It should also be noted that in the calculation of  $E_{\rm T}^{\rm miss, CellOut}$  the track momentum is used instead of the topocluster energy when there is a track-topocluster matching (see Section 5.3.1). This would result in a reduced uncertainty due to the more precise measurement of the track momentum, which is not taken into account here. Further study is expected to provide a reduction in this uncertainty in future, by considering the described effects in detail.

To give an estimate of the  $E_{\rm T}^{\rm miss, CellOut}$  systematic uncertainty, the calorimeter contribution can be taken from Section 7.2, and the uncertainty from the event generator settings from Section 7.1 (PYTHIA Perugia 2010 tune). This results in a total systematic uncertainty on the scale of  $E_{\rm T}^{\rm miss, CellOut}$  of about 13%, which slightly decreases when  $\sum E_{T}^{CellOut}$  increases.

#### 7.3 Evaluation of the systematic uncertainty on the $\boldsymbol{E}_{T}^{miss,softjets}$ scale

The same procedure described in the previous sections is used to assess the systematic uncertainty on the  $E_{\rm T}^{\rm miss}$  term calculated from soft jets (see Section 5.1).

Using the MC approach described in Section 7.1, it is found that the uncertainty on  $E_{\rm T}^{\rm miss, softjets}$  does not exhibit a large dependence on the event  $\Sigma E_{\rm T}$ , as was also found for the uncertainty on the  $E_{\rm T}^{\rm miss, CellOut}$  scale. The results are consistent between the QCD jet samples and the W samples, as can be seen from Table 3 which gives the systematic uncertainties  $R_i$ as computed in jet samples and in  $W \to \ell v$  samples.

A total, symmetric, systematic uncertainty of about 3.3% on the  $E_{\rm T}^{\rm miss, softjets}$  term is obtained by combining the results

Variation	jet events	W production	
Dead Material	$(-1.5\pm0.1)\%$	$(-1.5\pm0.2)\%$	
FTFP_BERT	$(0.3 \pm 0.4)\%$	$(0.8 \pm 0.2)\%$	
QGSP	$(-2.6\pm0.4)\%$	$(-2.5\pm0.2)\%$	
PYTHIA Perugia 2010 tune	$(-1.4\pm0.1)\%$	$(-1.0\pm0.2)\%$	

**Table 3.** Systematic uncertainties  $(R_i)$  on  $E_T^{\text{miss,softjets}}$  associated with variations in the dead material (all the variations listed in Table 1 are applied at the same time), in the calorimeter shower modelling (FTFP\_BERT, QGSP) and in the event generator settings (PYTHIA Perugia 2010 tune).

in Table 3, as was done in Section 7.1. With the same datadriven approach utilising the uncertainty on the topocluster energy scale described in Section 7.2, the systematic uncertainty on  $E_{\rm T}^{\rm miss, softjets}$  is evaluated to be about 10%.

As for  $E_{\rm T}^{\rm miss, CellOut}$ , the uncertainty on the  $E_{\rm T}^{\rm miss, softjets}$  scale found by shifting the topocluster energies is larger than the uncertainty estimated from MC simulation. To give an estimate of the systematic uncertainty on  $E_T^{\text{miss,softjets}}$ , the contribution from the calorimeter response can be taken from the datadriven evaluation and the contribution from the event generator settings from Table 3. This results in an overall systematic uncertainty of about 10% on  $E_{\rm T}^{\rm miss, softjets}$ , slightly increasing as  $\sum E_{\rm T}$  increases.

#### 7.4 Evaluation of the overall systematic uncertainty on the $E_{T}^{miss}$ scale in $W \rightarrow ev$ and $W \rightarrow \mu v$ events

Using as inputs the systematic uncertainties on the different re-constructed objects [8,21] and on  $E_{\rm T}^{\rm miss, CellOut}$  and  $E_{\rm T}^{\rm miss, softjets}$ evaluated in the previous sections, the overall  $E_{\rm T}^{\rm miss}$  systematic uncertainty in  $W \rightarrow ev$  and  $W \rightarrow \mu v$  events is estimated. The same method can be applied to any final state event topology. Figure 16 shows, for both  $W \rightarrow ev$  and  $W \rightarrow \mu v$  events, the systematic uncertainties on each of the terms  $E_T^{\text{miss},e}$  ( $E_T^{\text{miss},\mu}$ ),  $E_T^{\text{miss,jets}}$ ,  $E_T^{\text{miss,softjets}}$  and  $E_T^{\text{miss,CellOut}}$  as a function of their individual contribution to  $\sum E_{T}$  labelled  $\sum E_{T}^{\text{term}}$ . All the uncertainties are calculated with the formulae in Equations 9 and 10. In the same figure the uncertainty on  $E_{\rm T}^{\rm miss}$  due to the uncertainties on the different terms is also shown as a function of the total  $\sum E_{\rm T}$ , together with the overall uncertainty on  $E_{\rm T}^{\rm miss}$ , obtained by combining the partial terms. The uncertainties on  $E_{\rm T}^{\rm miss, softjets}$  and  $E_{\rm T}^{\rm miss, CellOut}$  are considered to be fully correlated. In  $W \rightarrow ev$  and  $W \rightarrow \mu v$  events, selected as described in Section 3.3, the overall uncertainty on the  $E_{T}^{miss}$  scale increases with  $\Sigma E_{\rm T}$  from ~ 1% to ~ 7%. It is estimated to be, on average, about 2.6% for both channels.

The  $E_{\rm T}^{\rm miss}$  scale uncertainty depends on the event topology because the contribution of a given  $E_{\rm T}^{\rm miss}$  term can vary for different final states.



**Fig. 16.** Fractional systematic uncertainty (calculated as in Equations 9 and 10) on different  $E_{\rm T}^{\rm miss}$  terms as a function of respective  $\sum E_{\rm T}^{\rm term}$  (left) and contributions of different term uncertainties on  $E_{\rm T}^{\rm miss}$  uncertainty as a function of  $\sum E_{\rm T}$  (right) in  $MCW \rightarrow ev$  events (top) and  $W \rightarrow \mu v$  events (bottom). The overall systematic uncertainty on the  $E_{\rm T}^{\rm miss}$  scale, obtained combining the various contributions is shown in the right plots (filled circles). The uncertainties on  $E_{\rm T}^{\rm miss, Softjets}$  and  $E_{\rm T}^{\rm miss, CellOut}$  are considered to be fully correlated.

# 8 Determination of the ${\pmb E}_T^{miss}$ scale from ${\pmb W} o \ell \nu$ events

scale. The lepton transverse momentum,  $p_T^{\ell}$ , and the  $E_T^{\text{miss}}$  are used to calculate  $m_T$  as:

$$m_T = \sqrt{2p_T^{\ell} E_T^{\text{miss}}(1 - \cos\phi)} \tag{11}$$

where  $\phi$  is the azimuthal angle between the lepton momentum and  $E_T^{\text{miss}}$  directions. The true  $m_T$  is reconstructed from the simulation under the hypothesis that  $E_T^{\text{miss}}$  is entirely due to the neutrino momentum,  $p_T^{\nu}$ . Template histograms of the  $m_T$  distributions are generated by convoluting the true transverse mass distribution with a Gaussian function:

$$E_{x(y)}^{\text{miss,smeared}} = \alpha E_{x(y)}^{\text{miss,True}} * Gauss(0, k \cdot \sqrt{\Sigma E_{\text{T}}})$$
(12)

where the parameters  $\alpha$  and k are the  $E_{\rm T}^{\rm miss}$  scale and resolution respectively.

The  $\alpha$  and k parameters are determined through a fit of the  $m_{\rm T}$  distribution to data using a linear combination of signal and background  $m_{\rm T}$  distributions obtained from simulation. All the backgrounds, with the exception of the jet background, are evaluated from the same MC samples used in Section 6.3 and the normalization is fixed according to their cross-sections. The

The determination of the absolute  $E_T^{\text{miss}}$  scale is important in a range of analyses involving  $E_T^{\text{miss}}$  measurements, ranging from precision measurements to searches for new physics.

In this section two complementary methods to determine the absolute scale of  $E_T^{\text{miss}}$  using  $W \to \ell v$  events are described. The first method uses a fit to the distribution of the transverse mass,  $m_T$ , of the lepton- $E_T^{\text{miss}}$  system, and is sensitive both to the scale and the resolution of  $E_T^{\text{miss}}$ . The second method uses the interdependence of the neutrino and lepton momenta in the  $W \to ev$  channel, and the  $E_T^{\text{miss}}$  scale is determined as a function of the reconstructed electron transverse momentum. Both methods allow checks on the agreement between data and MC simulation for the  $E_T^{\text{miss}}$  scale.

#### 8.1 Reconstructed transverse mass method

The method described in this section uses the shape of the  $m_{\rm T}$  distribution and is sensitive to both the  $E_{\rm T}^{\rm miss}$  resolution and

shape of the jet background is also evaluated from MC simulation and its normalization is obtained from the fit, in addition to  $\alpha$  and k.

To select  $W \to \mu v$  events, the same criteria as described in Section 3.3 are used, with the exception that no cut on  $E_{\rm T}^{\rm miss}$ is applied and a looser cut,  $m_{\rm T} > 30$  GeV, is applied in order that the background normalization can be fitted. The  $\alpha$  and kparameters obtained from the fit are shown in Table 4, together with the numbers of events for the signal and backgrounds and the  $\chi^2$ /ndof of the fit. In the table, instead of the values of  $\alpha$ , the values of  $\alpha - 1 = \langle (E_{x(y)}^{\rm miss,True}) / E_{x(y)}^{\rm miss,True} \rangle$  are reported, in order to compare with the result in Sections 6.3.1 and 8.2. The results for the  $\alpha$  and k parameters using the  $m_{\rm T}$  distribution of the simulated signal are also shown in Table 4, and they are in good agreement with the results from data. The result of the fit to data and MC simulation is shown in Figure 17.

To select  $W \rightarrow ev$  events, the selection described in Section 3.3 is used with the addition of tighter cuts. A cut  $E_{\rm T}^{\rm miss}$  > 36 GeV is applied to exclude the region where the  $E_{\rm T}^{\rm miss}$  response is not linear (see Figure 14). A cut  $m_{\rm T}$  > 40 GeV is also applied. The  $\alpha$  and k parameters obtained from the fit are shown in Table 4, together with the results obtained from the MC, which are in good agreement with data. The result of the fit to data and MC simulation is shown in Figure 17.

The results obtained with this method are compatible, at the few percent level, with the results shown in Figure 14 and Figure 15, which were derived using only simulation. From those figures, for the  $W \rightarrow \mu \nu$  channel  $\alpha - 1$  has values up to 3% and the resolution is  $0.47\sqrt{\Sigma E_T}$ ; for the  $W \rightarrow e\nu$  channel  $\alpha - 1$  is close to zero for high  $E_T^{\text{miss}}$  values and the resolution is  $0.47\sqrt{\Sigma E_T}$ .

The uncertainty due to background subtraction is already included in the uncertainty reported in Table 4. The systematic uncertainty on  $\alpha - 1$  is determined to be about 1% for each channel, by checking the stability of the results using different cuts on  $E_{\rm T}^{\rm miss}$  and using a different generator, MC@NLO. In summary, with this method the  $E_{\rm T}^{\rm miss}$  absolute scale is determined from  $W \to \ell v$  events, in a data sample corresponding to an integrated luminosity of about 36 pb<sup>-1</sup>, with an uncertainty (adding the uncertainties reported in Table 4 with the systematic uncertainty) of about 1.5% and about 2% for the  $W \to \mu v$  and  $W \to ev$  decay channels, respectively.

# 8.2 Method based on the correlation between electron and neutrino transverse momenta in $W \rightarrow ev$

In this section the correlation between the transverse momenta of charged and neutral leptons from W boson decays is used to determine the  $E_T^{\text{miss}}$  scale. The mean measured  $E_T^{\text{miss}}$  is compared to the mean true  $E_T^{\text{miss}}$  from signal MC events. The relative bias in the reconstructed  $E_T^{\text{miss}}$ ,  $(\langle E_T^{\text{miss}} \rangle - \langle E_T^{\text{miss}} \rangle T \rangle / \langle E_T^{\text{miss}} \rangle$ . Therefore the shape of the  $p_T^W$  distribution was distorted by  $\pm 10\%$ , justified by the comparison of a recent measurement of the  $p_T^W$  distributions, and the measurement of the  $p_T^W$  distribution, such as a function of  $p_T^e$  because the MC simulation of the electron response is more accurate than that for hadrons.

This method is shown for  $W \rightarrow ev$  events by applying selection criteria similar to the ones described in Section 3.3, but with isolation requirements both on the electron track and

calorimeter signal. The  $E_{\rm T}^{\rm miss}$  is required to be greater than 20 GeV and no cut is applied on  $m_{\rm T}$ .

MC samples are generated with MC@NLO [15]. A nextto-leading-order (NLO) generator is used for this study because in this approach the  $E_T^{\text{miss}}$  scale is validated on the basis of the known decay properties of the W boson. The correlation between  $p_T^v$  and  $p_T^e$  is important for this study, and is poorly described by leading-order generators such as PYTHIA, whereas it is much improved in MC@NLO. The MC events are weighted such that the true W boson transverse momentum,  $p_T^W$ , and pseudorapidity  $\eta^W$  agree with that generated using the RES-BOS [26] generator which is more accurate in describing  $p_T^W$  at low values. Finally, an additional smearing is applied to the reconstructed electron momentum in the MC samples, to match the electron resolution measured in data, and the correction is propagated to  $E_T^{\text{miss}}$ .

A data-driven technique is used to estimate the impact of jet background, which is small (see Figure 18 left) and concentrated at low  $p_T^e$ .  $W \to \tau v$  events, where the  $\tau$  decays to an electron, are the second largest background, but the impact on the mean value of  $E_T^{miss}$  is found to be negligible.

The distribution of  $p_T^e$  is shown in Figure 18. The distribution from data after event selection is fitted by varying the normalization of signal MC and QCD background distributions. A satisfactory description of data is achieved except for the first bin, which is excluded from the fit. For each  $p_T^e$  bin, the corrected distribution of  $E_T^{miss}$  is obtained by subtracting that of the background sample (after normalizing it according to the fit) from the data distribution. The largest impact of background corresponds to  $p_T^e = 20$  GeV, with an effect of about 2 GeV on the mean value of  $E_T^{miss}$ ; the effect decreases quickly to 0.2 GeV at  $p_T^e = 30$  GeV.

Since a cut on  $E_{\rm T}^{\rm miss}$  is used for the event selection and the  $E_{\rm T}^{\rm miss}$  resolution is finite, the results are biased. To correct for the bias in signal MC events the requirement of reconstructed  $E_{\rm T}^{\rm miss} > 20$  GeV is replaced by a cut on true  $E_{\rm T}^{\rm miss} > 20$  GeV.

The mean measured  $E_{\rm T}^{\rm miss}$ , corrected for background and for the event selection bias, is used to calculate the relative bias in the reconstructed  $E_{\rm T}^{\rm miss}$ ,  $(\langle E_{\rm T}^{\rm miss} \rangle - \langle E_{\rm T}^{\rm miss,{\rm True}} \rangle)/\langle E_{\rm T}^{\rm miss,{\rm True}} \rangle$ , which is shown in Figure 18 as a function of  $p_{\rm T}^e$ . The figure shows that the  $E_{\rm T}^{\rm miss}$  scale is correct at low values of  $p_{\rm T}^e$  while it is overestimated at high values of  $p_{\rm T}^e$ .

The bias on  $E_{\rm T}^{\rm miss}$  is on the percent level between 25 and 35 GeV, then it rises up to 7% and it is 2 +- 0.1% on average. For comparison, if the entire calculation is performed on signal MC events alone, the resulting average bias in  $E_{\rm T}^{\rm miss}$  is  $2.9 \pm 0.1\%$ . The method relies on simulation to derive the correlation between  $E_{\rm T}^{\rm miss,True}$  and  $p_{\rm T}^e$ , so it can be sensitive to details of the simulation. In particular, the jet factorization and renormalization scales, as well as the choice of PDF, can affect the results, but all these also change the  $p_{\rm T}^W$  distribution. **FTH** effore the shape of the  $p_{\rm T}^W$  distribution was distorted by  $\pm 10\%$ , justified by the comparison of a recent measurement of the  $p_{\rm T}^Z$  distribution [27] with RESBOS predictions, and the relative bias was calculated again. A systematic uncertainty on the relative  $E_{\rm T}^{\rm miss}$  scale are summarized in Table 5. These

Channel	α – 1 (%)	k	Signal	EW(fixed)	QCD	$\chi^2/ndof$
$W  ightarrow \mu v$ data	$5.1\pm0.8$	$0.52 \pm 0.01$	$164920 \pm 840$	14760	$24870 \pm 840$	68/87
$W \rightarrow \mu \nu MC$	$5.5\pm0.8$	$0.50\pm0.01$				70/78
$W \rightarrow e v$ data	$-0.8 \pm 1.6$	$0.49\pm0.01$	$75660 \pm 180$	1210	$980\pm180$	54/75
$W \rightarrow e \nu \text{ MC}$	$1.8\pm1.7$	$0.50 \pm 0.01$				38/54

**Table 4.** Results of  $m_T$  fit in  $W \to \ell v$  events. The second and third columns show the scale and resolution parameters obtained. The numbers of events for the signal, the electroweak and QCD backgrounds obtained from the fit are shown in the fourth, fifth and sixth columns for data. In the last column the  $\chi^2$ /ndof of the fit is reported. The errors are statistical and take into account background subtraction uncertainties and correlations.



**Fig. 17.** Distributions of the transverse mass,  $m_T$ , of the muon- $E_T^{miss}$  system (left) and of the electron- $E_T^{miss}$  system (right) for data. The  $m_T$  distributions from Monte Carlo simulation are superimposed, after each background sample is weighted as explained in the text. The main backgrounds are shown for  $W \to \mu \nu$ , the sum of all backgrounds is shown for  $W \to e\nu$ . The  $W \to \ell \nu$  MC signal histogram is obtained using the true  $E_T^{miss}$  smeared as in Equation (12) with the scale and resolution parameters obtained from the fit.



**Fig. 18.** Transverse momentum distribution of electron candidates in data, in signal MC with nominal event selection and with reversed cuts for background (QCD) from data (left). Relative bias in the reconstructed  $E_{\rm T}^{\rm miss}$ (right). Only statistical uncertainties are shown.

results agree within errors with the values of  $\alpha - 1$  shown in Table 4.

#### 9 Conclusion

The missing transverse momentum ( $E_T^{miss}$ ) has been measured in minimum bias, di-jet,  $Z \rightarrow \ell \ell$  and  $W \rightarrow \ell \nu$  events in 7 TeV *pp* collisions recorded with the ATLAS detector in 2010. The value of  $E_{\rm T}^{\rm miss}$  is reconstructed from calorimeter cells in topological clusters, with the exception of electrons and photons for which a different clustering algorithm is used, and from reconstructed muons. The cells are calibrated according to their parent particle type. The scheme yielding the best performance is evaluated to be that in which electrons are calibrated with the default electron calibration and photons are used at the EM scale, the  $\tau$ -jets and jets are calibrated with the local hadronic calibration (LCW), the jets with  $p_{\rm T}$  greater than 20 GeV are scaled to the jet energy scale, and the contri-

source	scale bias (%)
data	$2.0 \pm 0.1 \pm 2.0$
MC	$2.9\pm0.1$

**Table 5.** Average relative  $E_T^{\text{miss}}$  scale bias obtained from data and MC simulation from the electron-neutrino correlation method. The statistical and the systematic uncertainties are given for data.

bution from topoclusters not associated to high- $p_T$  objects is calculated with LCW calibration combined with tracking information.

Monte Carlo simulation is found to describe the data in general rather well. No large tails are observed in the  $E_{\rm T}^{\rm miss}$  distribution in minimum bias, di-jet and  $Z \rightarrow \ell \ell$  events, where no significant  $E_{\rm T}^{\rm miss}$  is expected. The tails are not completely well described by MC simulation especially in di-jets events, where there are more events in the tail in data.

There is some difference observed between data and MC simulation for the reconstructed total transverse energy. The precise difference is dependent on the model used to simulate soft-physics processes.

The  $E_{\rm T}^{\rm miss}$  resolution is similar in the different channels studied and in agreement with the resolution in the MC simulation. The resolution follows a function  $\sigma = k \cdot \sqrt{\Sigma E_{\rm T}}$ , where the parameter *k* is about 0.5 GeV<sup>1/2</sup>.

The linearity of the  $E_{\rm T}^{\rm miss}$  measurement in  $W \to \ell \nu$  events is studied in MC simulation as a function of the true  $E_{\rm T}^{\rm miss}$ . Except for the bias observed at small true  $E_{\rm T}^{\rm miss}$  values (visible up to 40 GeV), due to the finite  $E_{\rm T}^{\rm miss}$  resolution, the linearity is better than 1% in  $W \to e\nu$  events, while a small non-linearity up to about 3% is observed in  $W \to \mu\nu$  events.

The  $E_{\rm T}^{\rm miss}$  projected along the Z direction in  $Z \rightarrow \ell \ell$  events is observed to have a bias up to 6 GeV at large values of  $p_T^Z$ in events with no jets, suggesting that some improvements are still needed in the calibration of low- $p_{\rm T}$  objects.

The overall systematic uncertainty on  $E_{\rm T}^{\rm miss}$  scale, calculated by combining the uncertainties on the various terms entering the full  $E_{\rm T}^{\rm miss}$  calculation, is estimated to be, on average, 2.6% in events with a W decaying to a lepton (electron or muon) and neutrino. The uncertainty is larger at large  $\Sigma E_{\rm T}$ .

Two methods are used for determining the  $E_{\rm T}^{\rm miss}$  scale from  $W \rightarrow \ell \nu$  events in data, giving results in agreement with that evaluated using MC simulation. The resulting uncertainty on the  $E_{\rm T}^{\rm miss}$  scale determined in-situ with 36 pb<sup>-1</sup> of data is, on average, about 2%.

#### **10 Acknowledgements**

We thank CERN for the very successful operation of the LHC, as well as the support staff from our institutions without whom ATLAS could not be operated efficiently.

We acknowledge the support of ANPCyT, Argentina; Yer-PhI, Armenia; ARC, Australia; BMWF, Austria; ANAS, Azerbaijan; SSTC, Belarus; CNPq and FAPESP, Brazil; NSERC, NRC and CFI, Canada; CERN; CONICYT, Chile; CAS, MOST and NSFC, China; COLCIENCIAS, Colombia; MSMT CR, MPO CR and VSC CR, Czech Republic; DNRF, DNSRC and Lundbeck Foundation, Denmark; ARTEMIS, European Union; IN2P3-CNRS, CEA-DSM/IRFU, France; GNAS, Georgia; BMBF, DFG, HGF, MPG and AvH Foundation, Germany; GSRT, Greece; ISF, MINERVA, GIF, DIP and Benoziyo Center, Israel; INFN, Italy; MEXT and JSPS, Japan; CNRST, Morocco; FOM and NWO, Netherlands; RCN, Norway; MNiSW, Poland; GRICES and FCT, Portugal; MERYS (MECTS), Romania; MES of Russia and ROSATOM, Russian Federation; JINR; MSTD, Serbia; MSSR, Slovakia; ARRS and MVZT, Slovenia; DST/NRF, South Africa; MICINN, Spain; SRC and Wallenberg Foundation, Sweden; SER, SNSF and Cantons of Bern and Geneva, Switzerland; NSC, Taiwan; TAEK, Turkey; STFC, the Royal Society and Leverhulme Trust, United Kingdom; DOE and NSF, United States of America.

The crucial computing support from all WLCG partners is acknowledged gratefully, in particular from CERN and the AT-LAS Tier-1 facilities at TRIUMF (Canada), NDGF (Denmark, Norway, Sweden), CC-IN2P3 (France), KIT/GridKA (Germany), INFN-CNAF (Italy), NL-T1 (Netherlands), PIC (Spain), ASGC (Taiwan), RAL (UK) and BNL (USA) and in the Tier-2 facilities worldwide.

#### References

- 1. The ATLAS Collaboration, *The ATLAS Experiment at the CERN Large Hadron Collider*, JINST **3** (2008) S08003.
- 2. The ATLAS Collaboration, Performance of the ATLAS detector using first collision data, JHEP 09 (2010) 056.
- The ATLAS Collaboration, Search for squarks and gluinos using final states with jets and missing transverse momentum with the ATLAS detector in 7 TeV proton-proton collisions, Phys. Lett. B 701 (2011) 186.
- 4. The ATLAS Collaboration, *Properties of jets and inputs to jets reconstruction and calibration with the ATLAS detector using proton-proton collisions at*  $\sqrt{s} = 7$  *TeV*, ATLAS-CONF-2010-053.
- M. Cacciari, G. P. Salam, and G. Soyez, *The anti-k<sub>t</sub> jet clustering algorithm*, JHEP 05 (2008) 063, arXiv:0802.1189.
- The ATLAS Collaboration, Luminosity Determination in pp Collisions at 7 TeV using the ATLAS Detector at the LHC, EPJC 071 (2011) 1630, arXiv:1101.2185.
- The ATLAS Collaboration, Updated Luminosity Determination in pp Collisions at sqrt(s)=7 TeV using the ATLAS Detector, ATLAS-CONF-2011-011.
- 8. The ATLAS Collaboration, *Measurement of the*  $W \rightarrow lv$  and  $Z \rightarrow ll$  production cross sections in proton-proton collisions at  $\sqrt{s} = 7$  TeV with the ATLAS detector, JHEP **12** (2010) 060, arXiv:1010.2130.
- The ATLAS Collaboration, *Charged-particle multiplicities in pp* interactions measured with the ATLAS detector at the LHC, New J. Phys. 13 (2011) 053033.
- The ATLAS Collaboration, *Muon Reconstruction Performance*, ATLAS-CONF-2010-064.
- The ATLAS Collaboration, *Electron and photon reconstruction* and identification in ATLAS: expected performance at high energy and results at 900 GeV, ATLAS-CONF-2010-005.
- T. Sjostrand, S. Mrenna, and P. Skands, *PYTHIA 6.4 Physics and Manual*, JHEP 05 (2006) 026.
- The ATLAS Collaboration, Charged particle multiplicities in pp interactions at sqrt(s) = 0.9 and 7 TeV in a diffractive limited phase-space measured with the ATLAS detector at the LHC and new PYTHIA6 tune, ATLAS-CONF-2010-031.

- S. Agostinelli et al., *GEANT4: A simulation toolkit*, NIM A 506 (2003) 250.
- S. Frixione and B. R. Webber, *Matching NLO QCD* computations and parton shower simulations, JHEP 06 (2002) 029.
- T. Sjostrand, S. Mrenna, and P. Skands, Brief Introduction to PYTHIA8.1, Comput. Phys. Comm. 178 (2008) 852, arXiv:0710.3820.
- The ATLAS Collaboration, Expected Performance of the ATLAS Experiment - Detector, Trigger and Physics (Jet and E<sup>miss</sup><sub>T</sub> chapter), arXiv:0901.0512.
- 18. W. Lampl et al., *Calorimeter clustering algorithms: Description and performance*, ATL-LARG-PUB-2008-002.
- The ATLAS Collaboration, *Tau Reconstruction and Identification Performance in ATLAS*, ATLAS-CONF-2010-086.
   T. Barillari et al., *Local Hadron Calibration Properties*,
- ATL-LARG-PUB-2009-001.
- 21. The ATLAS Collaboration, Jet energy scale and its systematic uncertainty in proton-proton collisions at  $\sqrt{s} = 7$  TeV in ATLAS 2010 data, ATLAS-CONF-2011-032.
- 22. R. Corke and T. Sjostrand, Interleaved Parton Showers and Tuning Prospects, JHEP 032 (2011) 1103, arXiv:1011.1759.
- The ATLAS Collaboration, Measurement of inclusive jet and dijet cross sections in proton-proton collisions at 7 TeV centre-of-mass energy with the ATLAS detector, Eur. Phys. J. C 71 (2011) 1512, arXiv:1009.5908.
- 24. The ATLAS Collaboration, *Reconstruction and Calibration of Missing Transverse Energy and Performance in Z and W events in ATLAS Proton-Proton Collisions at*  $\sqrt{s} = 7$  TeV, ATLAS-CONF-2011-080.
- 25. The ATLAS Collaboration, *ATLAS Calorimeter Response to Single Isolated Hadrons and Estimation of the Calorimeter Jet Scale Uncertainty*, ATLAS-CONF-2011-028.
- C. Balazs and C. P. Yuan, Soft gluon effects on lepton pairs at hadron colliders, Phys. Rev. D 56 (1997) 55585583.
- 27. The ATLAS Collaboration, Measurement of the transverse momentum distribution of  $Z/\gamma$  bosons in proton-proton collisions at  $\sqrt{s} = 7$  TeV with the ATLAS detector, submitted to Phys. Lett. B, arXiv:1107.2381.

22

#### The ATLAS Collaboration

Da Anto-Schmanzander Permennance on Knosing Frankreike Konnennund Keisenkultum at y<sup>-12</sup> (197)
 S. Addina, S. Abdella, M. A. Abdelalim, "A. Abdeselam<sup>11,2</sup>, O. Abdinovi", B. Abl<sup>12</sup>, M. Abdilne<sup>45</sup>, H. Abrelli, "A. Abdesilam<sup>11,4</sup>, D. Abdams,", T. M. Addy, "J. Adelinan<sup>12</sup>, M. Abdella, "A. Ahaman, "A. Abdeselam<sup>11,4</sup>, O. Abdinovi,", J. Adelinan<sup>14</sup>, "A. Abdeselam<sup>11,4</sup>, O. Abdinovi, "A. Abdesilam<sup>14</sup>, "A. Abasan<sup>14</sup>, "A. Abasan<sup>14</sup>, "L. Akasan<sup>24</sup>, "A. Abdinovi, "A. Abdesilam<sup>14</sup>, "A. Abdeselam<sup>11,4</sup>, O. Abdinovi, "A. Abdeselam<sup>14,4</sup>, "A. Abasan<sup>14</sup>, "A. Abdeselam<sup>14,4</sup>, "A. Abasan<sup>14</sup>, "A. Abdeselam<sup>14,4</sup>, "A. Abasan<sup>14</sup>, "A. Abdeselam<sup>14,4</sup>, "A. Abasan<sup>14</sup>, "A. Abdeselam<sup>14,4</sup>, "A. Abdes

C.A. Chavez Barajas<sup>29</sup>, S. Cheatham<sup>85</sup>, S. Chekanov<sup>5</sup>, S.V. Chekulaev<sup>159a</sup>, G.A. Chelkov<sup>65</sup>, M.A. Chelstowska<sup>104</sup>, C. Chen<sup>64</sup>, H. Chen<sup>24</sup>, S. Chen<sup>32c</sup>, T. Chen<sup>32c</sup>, X. Chen<sup>172</sup>, S. Cheng<sup>32a</sup>, A. Cheplakov<sup>65</sup>, V.F. Chepurnov<sup>65</sup>, R. Cherkaoui El Moursli<sup>135e</sup>, V. Chernyatin<sup>24</sup>, E. Cheu<sup>6</sup>, S.L. Cheung<sup>158</sup>, L. Chevalier<sup>136</sup>, G. Chiefari<sup>102a,102b</sup>, L. Chikovani<sup>51</sup>, J.T. Childers<sup>58a</sup>, A. Chilingarov<sup>71</sup>, G. Chiodini<sup>72a</sup>, M.V. Chizhov<sup>65</sup>, G. Choudalakis<sup>30</sup>, S. Chouridou<sup>137</sup>, I.A. Christidi<sup>77</sup>, A. Christov<sup>48</sup>, D. Chromek-Burckhart<sup>29</sup>, M.L. Chu<sup>151</sup>, J. Chudoba<sup>125</sup>, G. Ciapetti<sup>132a,132b</sup>, K. Ciba<sup>37</sup>, A.K. Ciftci<sup>3a</sup>, R. Ciftci<sup>3a</sup>, D. Cinca<sup>33</sup>, V. Cindro<sup>74</sup>, M.D. Ciobotaru<sup>163</sup>, C. Ciocca<sup>19a,19b</sup>, A. Ciocio<sup>14</sup>, M. Cirilli<sup>87</sup>, M. Ciubancan<sup>25a</sup>, A. Clark<sup>49</sup>, P.J. Clark<sup>45</sup>, W. Cleland<sup>123</sup>, J.C. Clemens<sup>83</sup>, B. Clement<sup>55</sup>, C. Clement<sup>146a,146b</sup>, R.W. Clifft<sup>129</sup>, Y. Coadou<sup>83</sup>, M. Cobal<sup>164a,164c</sup>, A. Coccaro<sup>50a,50b</sup>, J. Cochran<sup>64</sup>, P. Coe<sup>118</sup>, J.G. Cogan<sup>143</sup>, J. Coggeshall<sup>165</sup>, E. Cogneras<sup>177</sup>, C.D. Cojocaru<sup>28</sup>, J. Colas<sup>4</sup>, A. P. Collin<sup>105</sup>, C. Collard<sup>115</sup>, N.L. Collins<sup>17</sup>, C. Collins-Tooth<sup>53</sup>, L. Colan<sup>55</sup>, G. Colon<sup>84</sup>, P. Conde Muiño<sup>124a</sup>, F. Conjavitis<sup>118</sup> A. Coccaro<sup>105,1</sup>, J. Cochran<sup>1,1</sup>, P. Coe<sup>10,2</sup>, J.G. Cogan<sup>1,4</sup>, J. Coggeshal<sup>10,5</sup>, E. Cogneras<sup>1,4</sup>, C.D. Cojocaru<sup>10,5</sup>, J. Colas<sup>1,5</sup>, A.P. Colijn<sup>105</sup>, C. Collard<sup>115</sup>, N.J. Collins<sup>17</sup>, C. Collins-Tooth<sup>53</sup>, J. Collot<sup>55</sup>, G. Colon<sup>84</sup>, P. Conde Muiño<sup>124a</sup>, E. Coniavitis<sup>118</sup>, M.C. Conidi<sup>11</sup>, M. Consonni<sup>104</sup>, V. Consorti<sup>48</sup>, S. Constantinescu<sup>25a</sup>, C. Conta<sup>119a,119b</sup>, F. Conventi<sup>102a,i</sup>, J. Cook<sup>29</sup>, M. Cooke<sup>14</sup>, B.D. Cooper<sup>77</sup>, A.M. Cooper-Sarkar<sup>118</sup>, N.J. Cooper-Smith<sup>76</sup>, K. Copic<sup>34</sup>, T. Cornelissen<sup>50a,50b</sup>, M. Corradi<sup>19a</sup>, F. Corriveau<sup>85,j</sup>, A. Cortes-Gonzalez<sup>165</sup>, G. Cortiana<sup>99</sup>, G. Costa<sup>89a</sup>, M.J. Costa<sup>167</sup>, D. Costanzo<sup>139</sup>, T. Costin<sup>30</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, D. Côté<sup>29</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>111</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>111</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>110</sup>, Costa<sup>111</sup>, Costa<sup>110</sup>, Costa<sup>111</sup>, Costa<sup>110</sup>, Costa<sup>111</sup>, F. Corriveau<sup>65,7</sup>, A. Cortes-Gonzalez<sup>105</sup>, G. Cortiana<sup>75</sup>, G. Costa<sup>105,4</sup>, M.J. Costa<sup>105</sup>, D. Costanzo<sup>105</sup>, I. Costin<sup>50</sup>, D. Costa<sup>105,7</sup>, A. Cortes-Gonzalez<sup>105</sup>, G. Cortiana<sup>75</sup>, C. Cowden<sup>27</sup>, B.E. Cox<sup>82</sup>, K. Cranmer<sup>108</sup>, F. Crescioli<sup>122a,122b</sup>, M. Cristinziani<sup>20</sup>, G. Crosetti<sup>36a,36b</sup>, R. Crupi<sup>72a,72b</sup>, S. Crépé-Renaudin<sup>55</sup>, C.-M. Cuciuc<sup>25a</sup>, C. Cuenca Almenar<sup>175</sup>, T. Cuhadar Donszelmann<sup>139</sup>, M. Curatolo<sup>47</sup>, C.J. Curtis<sup>17</sup>, P. Cwetanski<sup>61</sup>, H. Czirr<sup>141</sup>, Z. Czyczula<sup>117</sup>, S. D'Auria<sup>53</sup>, M. D'Onofrio<sup>73</sup>, A. D'Orazio<sup>132a,132b</sup>, P.V.M. Da Silva<sup>23a</sup>, C. Da Via<sup>82</sup>, W. Dabrowski<sup>37</sup>, T. Dai<sup>87</sup>, C. Dallapiccola<sup>84</sup>, M. D'Onofrio<sup>73</sup>, A. D'Orazio<sup>132a,132b</sup>, P.V.M. Da Silva<sup>23a</sup>, C. Da Via<sup>82</sup>, W. Dabrowski<sup>37</sup>, T. Dai<sup>87</sup>, C. Dallapiccola<sup>84</sup>, C. Chadar Donszelmann<sup>19</sup>, M. Curatolo<sup>47</sup>, C.J. Curiis<sup>17</sup>, P. Cvettanski<sup>61</sup>, H. Czin<sup>141</sup>, Z. Czyzual<sup>31</sup>, S. D. Yania<sup>53</sup>, M. Damers<sup>167</sup>, A. D. Gardis<sup>126,135</sup>, P. M. Dasives<sup>167</sup>, L. Dakroski<sup>77</sup>, T. Dai<sup>87</sup>, C. Dallapiccola<sup>84</sup>, M. Damis<sup>157</sup>, H.O. Danielsson<sup>27</sup>, D. Dannheim<sup>99</sup>, V. Dao<sup>99</sup>, G. Darbo<sup>50</sup>, G.L. Darlea<sup>259</sup>, C. Daum<sup>165</sup>, J. P. Daverg<sup>367</sup>, B. Damiani<sup>157</sup>, H.O. Danielsson<sup>27</sup>, D. Dannheim<sup>99</sup>, V. Dao<sup>99</sup>, G. Darbo<sup>50</sup>, G.L. Darlea<sup>259</sup>, C. Daum<sup>165</sup>, J. P. Daverg<sup>367</sup>, E. Dawe<sup>176</sup>, I. Dawson<sup>157</sup>, J. W. Dawson<sup>5</sup>, R.K. Day<sup>29</sup>, K. De<sup>7</sup>, R. de Asmundis<sup>1020</sup>, S. De Castro Faria Salgado<sup>21</sup>, S. De Cecco<sup>73</sup>, J. de Graat<sup>28</sup>, N. De Groot<sup>110</sup>, P. de Jong<sup>103</sup>, C. De La Taill<sup>111</sup>, H. De la Tore<sup>80</sup>, B. De Lotto<sup>164,164</sup>, L. De Wono<sup>111</sup>, D. De Vive Te Regie<sup>1125</sup>, S. Dean<sup>77</sup>, D. De Pedis<sup>1253</sup>, A. De Salvo<sup>123</sup>, J. De Salvo<sup>124</sup>, J. De Salvo<sup>126</sup>, J. Degenhand<sup>113</sup>, M. Delha<sup>118</sup>, C. Del Papal<sup>654,164</sup>, J. Del Petis<sup>1251,228</sup>, M. Deliyergiyev<sup>74</sup>, A. Dell'Acqu<sup>23</sup>, I. Dell'Acqu<sup>235</sup>, S. Demer<sup>78</sup>, P. Derra<sup>75</sup>, K. Desch<sup>176</sup>, E. Devetak<sup>148</sup>, P.O. Deviveiros<sup>158</sup>, P. Denisov<sup>128</sup>, D. Derendar<sup>128</sup>, J. Dell's, S. Dani<sup>77</sup>, D. Dell's Cu<sup>1251,129</sup>, M. Dell's Peti<sup>1251</sup>, D. Deres<sup>176</sup>, A. Dell's <sup>1261,129</sup>, M. Dell's <sup>1261,129</sup>, M. Dell's <sup>1261,129</sup>, A. D. Dia<sup>1261,130</sup>, A. D. Bill'and<sup>1251,130</sup>, A. D. Giaccio<sup>1354,130</sup>, A. D. Bill'and<sup>1254,130</sup>, A. D. Dokl<sup>254,150</sup>, D. Dol<sup>254,250</sup>, D. Dol<sup>254,250</sup>, D. Dol<sup>254,250</sup>, D. Dia<sup>254,250</sup>, D. Dol<sup>254,250</sup>, D. Dol<sup>254,250</sup>, D M. Dam<sup>35</sup>, M. Dameri<sup>50a,50b</sup>, D.S. Damiani<sup>137</sup>, H.O. Danielsson<sup>29</sup>, D. Dannheim<sup>99</sup>, V. Dao<sup>49</sup>, G. Darbo<sup>50a</sup>, G.L. Darlea<sup>25b</sup>,

25

C. Gemme<sup>50a</sup>, A. Gemmell<sup>53</sup>, M.H. Genest<sup>98</sup>, S. Gentile<sup>132a,132b</sup>, M. George<sup>54</sup>, S. George<sup>76</sup>, P. Gerlach<sup>174</sup>, A. Gershon<sup>153</sup>, C. Geweniger<sup>58a</sup>, H. Ghazlane<sup>135b</sup>, P. Ghez<sup>4</sup>, N. Ghodbane<sup>33</sup>, B. Giacobbe<sup>19a</sup>, S. Giagu<sup>132a,132b</sup>, V. Giakoumopoulou<sup>8</sup>, V. Giangiobbe<sup>122a,122b</sup>, F. Gianotti<sup>29</sup>, B. Gibbard<sup>24</sup>, A. Gibson<sup>158</sup>, S.M. Gibson<sup>29</sup>, L.M. Gilbert<sup>118</sup>, M. Gilchriese<sup>14</sup>, V. Gilewsky<sup>91</sup>, D. Gillberg<sup>28</sup>, A.R. Gillman<sup>129</sup>, D.M. Gingrich<sup>2,e</sup>, J. Ginzburg<sup>153</sup>, N. Giokaris<sup>8</sup>, R. Giordano<sup>102a,102b</sup>, F.M. Giorgi<sup>15</sup>, P. Giovannini<sup>99</sup>, P.F. Giraud<sup>136</sup>, D. Giugni<sup>89a</sup>, M. Giunta<sup>132a,132b</sup>, P. Giusti<sup>19a</sup>, B.K. Gjelsten<sup>117</sup>, L.K. Gladilin<sup>97</sup>, C. Glasman<sup>80</sup>, J. Glatzer<sup>48</sup>, A. Glazov<sup>41</sup>, K.W. Glitza<sup>174</sup>, G.L. Glonti<sup>65</sup>, J. Godfrey<sup>142</sup>, J. Godlewski<sup>29</sup>, M. Goebel<sup>41</sup>, T. Göpfert<sup>43</sup>, C. Goeringer<sup>81</sup>, C. Gössling<sup>42</sup>, T. Göttfert<sup>99</sup>, S. Goldfarb<sup>87</sup>, D. Goldin<sup>39</sup>, T. Golling<sup>175</sup>, S.N. Golovnia<sup>128</sup>, A. Gomes<sup>124a,b</sup>, L.S. Gomez Fajardo<sup>41</sup>, R. Gonçalo<sup>76</sup>, J. Goncalves Pinto Firmino Da Costa<sup>41</sup>, L. Gonella<sup>20</sup>, A. Gonidec<sup>29</sup>, A. Gonies <sup>1,2</sup>, E.S. Goniez Fajardo<sup>-</sup>, K. Goniçaro<sup>-</sup>, J. Gonicaros Finto Fintino Da Costa<sup>-</sup>, L. Goniera<sup>-</sup>, A. Gonidec<sup>-</sup>,
 S. Gonzalez<sup>172</sup>, S. González de la Hoz<sup>167</sup>, M.L. Gonzalez Silva<sup>26</sup>, S. Gonzalez-Sevilla<sup>49</sup>, J.J. Goodson<sup>148</sup>, L. Goossens<sup>29</sup>,
 P.A. Gorbounov<sup>95</sup>, H.A. Gordon<sup>24</sup>, I. Gorelov<sup>103</sup>, G. Gorfine<sup>174</sup>, B. Gorini<sup>29</sup>, E. Gorini<sup>72a,72b</sup>, A. Gorišek<sup>74</sup>, E. Gornicki<sup>38</sup>,
 S.A. Gorokhov<sup>128</sup>, V.N. Goryachev<sup>128</sup>, B. Gosdzik<sup>41</sup>, M. Gosselink<sup>105</sup>, M.I. Gostkin<sup>65</sup>, I. Gough Eschrich<sup>163</sup>, M. Gouighri<sup>135a</sup>, P.A. Gordounov<sup>29</sup>, H.A. Gordon<sup>4,1</sup>, I. Gorelov<sup>11,3</sup>, G. Gorfine<sup>1/4</sup>, B. Gorini<sup>2/2</sup>, L. Gorisk<sup>4,4</sup>, E. Gorrick<sup>1/3</sup>, S. Gorszk<sup>1/4</sup>, M. Gosselink<sup>10,5</sup>, M. Gossk<sup>11/6,5</sup>, M. Gongh Eschrich<sup>10,5</sup>, M. Gong<sup>11/13,6</sup>, D. Goujdami <sup>135,5</sup>, M.P. Goulette<sup>40</sup>, A.G. Goussiou<sup>13,8</sup>, C. Goy<sup>4</sup>, I. Grabowska-Bold<sup>16,5</sup>, V. Grask<sup>11/6</sup>, M. Gargy<sup>29</sup>, J.A. Grag<sup>14,4</sup>, E. Graztani<sup>134,4</sup>, O.G. Grebenyuk<sup>121</sup>, D. Greenfield<sup>129</sup>, T. Greenshaw<sup>13,3</sup>, Z.D. Greenwood<sup>24,1</sup>, K. Gregersen<sup>35</sup>, I.M. Gregor<sup>14</sup>, P. Grenier<sup>143</sup>, J. Griffith, <sup>136</sup>, N. Grigatashvill<sup>60</sup>, A.A. Grillo<sup>137</sup>, S. Grinstein<sup>11</sup>, Y. U. Grishkevich<sup>97</sup>, J. F. Grivau<sup>17,4</sup>, J. Grosse-Knette<sup>47,4</sup>, J. Gortinete<sup>12,5</sup>, D. Guest<sup>11,5</sup>, C. Guicheney<sup>31</sup>, A. Guida<sup>12,2,129</sup>, T. Guillemin<sup>4</sup>, S. Guindon<sup>4,4</sup>, H. Guler<sup>58,4</sup>, J. Gunthel<sup>12,5</sup>, D. Guest<sup>11,5</sup>, C. Guicheney<sup>31</sup>, A. Guida<sup>12,2,129</sup>, T. Guillemin<sup>4</sup>, S. Guindon<sup>4,4</sup>, H. Guler<sup>58,4</sup>, J. Gunthel<sup>12,5</sup>, D. Guy<sup>11,5</sup>, C. Guida<sup>11,5</sup>, A. Guirce<sup>21</sup>, P. Guirce<sup>21,11</sup>, N. Guttami<sup>15,5</sup>, O. Gutzwill<sup>11,6</sup>, A. Gugui<sup>15,6</sup>, Y. Gussko<sup>16,4</sup>, J. Haars<sup>20,5</sup>, T. Hans<sup>20,5</sup>, J. Haar<sup>20,5</sup>, J. Buka<sup>11,5</sup>, A. Hans<sup>20,5</sup>, T. Haar<sup>20,5</sup>, S. Hasse<sup>20,5</sup>, J. B. Hansen<sup>35</sup>, J.B. Hansen<sup>35</sup>, P. Hahn<sup>29,5</sup>, S. Haide<sup>29,5</sup>, X. Haar<sup>10,6</sup>, O. A. Hare<sup>15,7</sup>, T. Harenberg<sup>17,4</sup>, S. Hafakusha<sup>20,5</sup>, D. Harge<sup>27,4</sup>, S. Hasse<sup>10,4</sup>, J. Hatse<sup>13,4</sup>, K. Hassi<sup>10,4</sup>, M. Hanse<sup>15,4</sup>, T. Hatse<sup>16,4</sup>, T. Hatse<sup>16,5</sup>, T. Hausso<sup>11,5</sup>, J. Hatt<sup>24,5</sup>, S. Hausse<sup>14,5</sup>, J. Hausso<sup>14,5</sup>, J. Ha D. Goujdami<sup>135c</sup>, M.P. Goulette<sup>49</sup>, A.G. Goussiou<sup>138</sup>, C. Goy<sup>4</sup>, I. Grabowska-Bold<sup>163,g</sup>, V. Grabski<sup>176</sup>, P. Grafström<sup>29</sup>, K. Kawagoe<sup>07</sup>, T. Kawamoto<sup>135</sup>, G. Kawamura<sup>81</sup>, M.S. Kayl<sup>105</sup>, V.A. Kazanin<sup>107</sup>, M.Y. Kazarinov<sup>65</sup>, J.R. Keates<sup>82</sup>, R. Keeler<sup>169</sup>, R. Kehoe<sup>39</sup>, M. Keil<sup>54</sup>, G.D. Kekelidze<sup>65</sup>, M. Kelly<sup>82</sup>, J. Kennedy<sup>98</sup>, C.J. Kenney<sup>143</sup>, M. Kenyon<sup>53</sup>, O. Kepka<sup>125</sup>, N. Kerschen<sup>29</sup>, B.P. Kerševan<sup>74</sup>, S. Kersten<sup>174</sup>, K. Kessoku<sup>155</sup>, C. Ketterer<sup>48</sup>, J. Keung<sup>158</sup>, M. Khakzad<sup>28</sup>, F. Khalil-zada<sup>10</sup>, H. Khandanyan<sup>165</sup>, A. Khanov<sup>112</sup>, D. Kharchenko<sup>65</sup>, A. Khodinov<sup>96</sup>, A.G. Kholodenko<sup>128</sup>, A. Khomich<sup>58a</sup>, T.J. Khoo<sup>27</sup>, G. Khoriauli<sup>20</sup>, A. Khoroshilov<sup>174</sup>, N. Khovanskiy<sup>65</sup>, V. Khovanskiy<sup>95</sup>, E. Khramov<sup>65</sup>, J. Khubua<sup>51</sup>, H. Kim<sup>7</sup>, M.S. Kim<sup>2</sup>, P.C. Kim<sup>143</sup>, S.H. Kim<sup>160</sup>, N. Kimura<sup>170</sup>, O. Kind<sup>15</sup>, B.T. King<sup>73</sup>, M. King<sup>67</sup>, R.S.B. King<sup>118</sup>, J. Kirk<sup>129</sup>, G.P. Kirsch<sup>118</sup>, L.E. Kirsch<sup>22</sup>, A.E. Kiryunin<sup>99</sup>, T. Kishimoto<sup>67</sup>, D. Kisielewska<sup>37</sup>, T. Kittelmann<sup>123</sup>, A.M. Kiver<sup>128</sup>, H. Kiyamura<sup>67</sup>, E. Kladiva<sup>144b</sup>, J. Klaiber-Lodewigs<sup>42</sup>, M. Klein<sup>73</sup>, U. Klein<sup>73</sup>, K. Kleinknecht<sup>81</sup>, M. Klemetti<sup>85</sup>, A. Klier<sup>171</sup>, A. Klimentov<sup>24</sup>,

R. Klingenberg<sup>42</sup>, E.B. Klinkby<sup>35</sup>, T. Klioutchnikova<sup>29</sup>, P.F. Klok<sup>104</sup>, S. Klous<sup>105</sup>, E.-E. Kluge<sup>58a</sup>, T. Kluge<sup>73</sup>, P. Kluit<sup>105</sup>, S. Kluth<sup>99</sup>, N.S. Knecht<sup>158</sup>, E. Kneringer<sup>62</sup>, J. Knobloch<sup>29</sup>, E.B.F.G. Knoops<sup>83</sup>, A. Knue<sup>54</sup>, B.R. Ko<sup>44</sup>, T. Kobayashi<sup>155</sup>, M. Kobel<sup>43</sup>, M. Kocian<sup>143</sup>, A. Kocnar<sup>113</sup>, P. Kodys<sup>126</sup>, K. Köneke<sup>29</sup>, A.C. König<sup>104</sup>, S. Koenig<sup>81</sup>, L. Köpke<sup>81</sup>, F. Koetsveld<sup>104</sup>, P. Koevesarki<sup>20</sup>, T. Koffas<sup>29</sup>, E. Koffeman<sup>105</sup>, F. Kohn<sup>54</sup>, Z. Kohout<sup>127</sup>, T. Kohriki<sup>66</sup>, T. Koi<sup>143</sup>, T. Kokott<sup>20</sup>, G.M. Kolachev<sup>107</sup>, H. Kolanoski<sup>15</sup>, V. Kolesnikov<sup>65</sup>, I. Koletsou<sup>89a</sup>, J. Koll<sup>88</sup>, D. Kollar<sup>29</sup>, M. Kollefrahf<sup>4</sup>, S.D. Kolya<sup>82</sup>, A.A. Komar<sup>94</sup>, J.R. Komzagiri<sup>142</sup>, Y. Komori<sup>155</sup>, T. Kondo<sup>66</sup>, T. Kono<sup>41,o</sup>, A.I. Kononov<sup>48</sup>, R. Konoplich<sup>108,p</sup>, N. Konstantinidis<sup>77</sup>, A. Kotorkov<sup>128</sup>, K. Korcyl<sup>38</sup>, K. Korcyl<sup>38</sup>, K. Koreshev<sup>128</sup>, A. Korn<sup>14</sup>, A. Korol<sup>107</sup>, I. Korolkov<sup>11</sup>, E.V. Korolkova<sup>139</sup>, V.A. Korotkov<sup>128</sup>, K. Korcyl<sup>38</sup>, K. Kordasl<sup>54</sup>, V. Korshevu<sup>128</sup>, A. Korol<sup>107</sup>, I. Korolkov<sup>11</sup>, E.V. Korolkova<sup>139</sup>, V.A. Korukounelis<sup>8</sup>, V. Kouskoura<sup>154</sup>, A. Koutsman<sup>105</sup>, R. Kowalewski<sup>169</sup>, T.Z. Kowalski<sup>37</sup>, W. Kozanecki<sup>136</sup>, A.S. Kozhin<sup>128</sup>, V. Kral<sup>127</sup>, V.A. Kraamarenko<sup>97</sup>, G. Kramberger<sup>74</sup>, M.W. Krasny<sup>78</sup>, A. Krasznahorkay<sup>108</sup>, J. Kraus<sup>88</sup>, A. Kreisel<sup>153</sup>, F. Krejci<sup>127</sup>, J. Kretzschmar<sup>73</sup>, N. Krieger<sup>54</sup>, P. Krieger<sup>158</sup>, K. Kroeninger<sup>54</sup>, H. Krola<sup>99</sup>, J. Kroll<sup>20</sup>, J. Krosbeberg<sup>20</sup>, J. Krstic<sup>12a</sup>, U. Kruchonak<sup>65</sup>, H. Kriger<sup>20</sup>, T. Kruker<sup>16</sup>, Z.V. Krumshteyn<sup>65</sup>, A. Kuth<sup>20</sup>, T. Kubota<sup>86</sup>, S. Kuehn<sup>48</sup>, A. Kugel<sup>58e</sup>, T. Kuhl<sup>41</sup>, D. Kuhn<sup>62</sup>, V. Kukhtin<sup>65</sup>, Y. Kulchitsky<sup>90</sup>, S. Kuleshov<sup>31b</sup>, C. Kummer<sup>98</sup>, M. Kuna<sup>78</sup>, N. Kundu<sup>118</sup>, J. Kunkle<sup>120</sup>, A. Kupco<sup>125</sup>, H. Kurashige<sup>67</sup>, M. Kurata<sup>160</sup>, Y.A. Kurochkin<sup>90</sup>, V. Kus<sup>125</sup>, W. Kuykendall<sup>138</sup>, K. Kuze<sup>157</sup>, F. Laavat<sup>132</sup>, L. Laake<sup>15</sup>, D. Laaour<sup>78</sup>, V. K. Laantona<sup>56</sup>, K. Laahgu<sup>55</sup>, T. Laahgu<sup>78</sup>, S. Lai<sup>48</sup>, E. Laatosn<sup>55</sup>, M. Lamanna<sup>29</sup>, C. Laampe<sup>79</sup>, M. Latosn<sup>29</sup>, S. R. Klingenberg<sup>42</sup>, E.B. Klinkby<sup>35</sup>, T. Klioutchnikova<sup>29</sup>, P.F. Klok<sup>104</sup>, S. Klous<sup>105</sup>, E.-E. Kluge<sup>58a</sup>, T. Kluge<sup>73</sup>, P. Kluit<sup>105</sup>, A. Leger<sup>44</sup>, B.C. LeGeyl<sup>120</sup>, F. Legger<sup>98</sup>, C. Leggett<sup>44</sup>, M. Lehmacher<sup>20</sup>, G. Lehmann Miotto<sup>29</sup>, X. Lei<sup>6</sup>, M.A.L. Leite<sup>13ad</sup>, R. Leitner<sup>12ad</sup>, D. Lellouch<sup>171</sup>, M. Leltchouk<sup>34</sup>, B. Lemme<sup>54</sup>, V. Lendermann<sup>55a</sup>, K.J.C. Leneyl<sup>145</sup>), T. Lenz<sup>105</sup>, G. Lenzen<sup>174</sup>, B. Lenzi<sup>29</sup>, K. Leonhard<sup>143</sup>, S. Leontsins<sup>8</sup>, C. Leroy<sup>93</sup>, J.R. Lesser<sup>146a</sup>, C.G. Lester<sup>27</sup>, A. Leung Fook Cheong<sup>172</sup>, J. Levton<sup>87</sup>, L.J. Levinson<sup>171</sup>, M.S. Levitski<sup>128</sup>, M. Lewandowska<sup>21</sup>, A. Lewis<sup>118</sup>, G.H. Lewis<sup>108</sup>, A.M. Leyko<sup>20</sup>, M. Leyton<sup>15</sup>, B. Li<sup>31</sup>, R. Lifshti<sup>125</sup>, J.N. Lilley<sup>17</sup>, C. Limbach<sup>20</sup>, A. Linosani<sup>16</sup>, M. Limper<sup>63</sup>, S.C. Lind<sup>151</sup>, F. Linde<sup>105</sup>, J.T. Linnemann<sup>88</sup>, E. Lipeles<sup>120</sup>, L. Lipinsky<sup>125</sup>, A. Lipniacka<sup>13</sup>, T.M. Liss<sup>165</sup>, D. Lissauer<sup>44</sup>, A. Lister<sup>49</sup>, A.M. Litke<sup>137</sup>, C. Liu<sup>23</sup>, D. Liu<sup>151, \*</sup>, H. Liu<sup>87</sup>, J.B. Liu<sup>87</sup>, M. Liu<sup>230</sup>, S. Liu<sup>2</sup>, Y. Liu<sup>230</sup>, M. Liva<sup>1190</sup>, S.S.A. Livermore<sup>118</sup>, A. Leres<sup>55</sup>, J. Llovert<sup>85</sup>, E. Lobodzinsk<sup>41</sup>, P. Loch<sup>65</sup>, W.S. Lockman<sup>137</sup>, S. Lockwit<sup>175</sup>, T. Loddenkoetter<sup>20</sup>, F.K. Loebinger<sup>82</sup>, A. Loginov<sup>175</sup>, C.W. Loh<sup>168</sup>, T. Lohse<sup>15</sup>, K. Lohwasser<sup>48</sup>, M. Lokajicek<sup>125</sup>, J. Loken <sup>118</sup>, V.P. Lombardo<sup>4</sup>, R.E. Long<sup>71</sup>, L. Lopes<sup>124a,b</sup>, D. Lopez Mateos<sup>37</sup>, M. Losada<sup>162</sup>, P. Loscutoff<sup>14</sup>, F. Lo Sterzo<sup>132a,132b</sup>, M.J. Losty<sup>159a</sup>, X. Luo<sup>40</sup>, A. Louwi<sup>43</sup>, D. Ludwig<sup>44</sup>, D. Ludwig<sup>44</sup>, J. Ludwig<sup>48</sup>, J. Ludwig<sup>48</sup>, F. Luehring<sup>61</sup>, G. Luij<sup>2x,150</sup>, G. Luu<sup>29</sup>, D. Luot<sup>457</sup>, K. Lucotte<sup>55</sup>, A. Ludwig<sup>43</sup>, D. Ludwig<sup>44</sup>, J. Ludwig<sup>44</sup>, F. Luehring<sup>61</sup>, G. Luij<sup>2x,150</sup>, T. Maeino<sup>174</sup>, B. Sundiacthi<sup>274,150</sup>, C. Macian Chi<sup>274,150</sup>, J. Ludwig<sup>45</sup>, M. Macian Goia<sup>29</sup>, G. Maccarrone<sup>47</sup>, A. Macchiolo<sup>99</sup>, B. Maček<sup>74</sup>, J. Maciantohi<sup>275,150</sup>, L. Mal<sup>476,160,150</sup>, J. Lumb<sup>487,150</sup>, C. Mater<sup>457</sup>, J. Ludwi<sup>443</sup>, R. Maenner<sup>586</sup>, T. Maeno<sup>34</sup>, P. Matitg<sup>174</sup>, S. Maittig<sup>174</sup>, J. Madiatchi<sup>175,150</sup>, M. Martia<sup>574,150</sup>, C. Matarit<sup>574,150</sup>, C. Matarit<sup>574,150</sup>, M. Matita<sup>574,150</sup>, M. Matita<sup>574,150</sup>, M. Matek<sup>138</sup>, P. Matita<sup>574,150</sup>, M. Maten R.A. McLaren<sup>29</sup>, T. Mclaughlan<sup>17</sup>, S.J. McMahon<sup>129</sup>, R.A. McPherson<sup>109,7</sup>, A. Meade<sup>34</sup>, J. Mechnich<sup>105</sup>, M. Mechtel<sup>174</sup>,
M. Medinnis<sup>41</sup>, R. Meera-Lebbai<sup>111</sup>, T. Meguro<sup>116</sup>, R. Mehdiyev<sup>93</sup>, S. Mehlhase<sup>35</sup>, A. Mehta<sup>73</sup>, K. Meier<sup>58a</sup>, J. Meinhardt<sup>48</sup>,
B. Meirose<sup>79</sup>, C. Melachrinos<sup>30</sup>, B.R. Mellado Garcia<sup>172</sup>, L. Mendoza Navas<sup>162</sup>, Z. Meng<sup>151,s</sup>, A. Mengarelli<sup>19a,19b</sup>,
S. Menke<sup>99</sup>, C. Menot<sup>29</sup>, E. Meoni<sup>11</sup>, K.M. Mercurio<sup>57</sup>, P. Mermod<sup>118</sup>, L. Merola<sup>102a,102b</sup>, C. Meroni<sup>89a</sup>, F.S. Merritt<sup>30</sup>,
A. Messina<sup>29</sup>, J. Metcalfe<sup>103</sup>, A.S. Mete<sup>64</sup>, S. Meuser<sup>20</sup>, C. Meyer<sup>81</sup>, J-P. Meyer<sup>136</sup>, J. Meyer<sup>173</sup>, J. Meyer<sup>54</sup>, T.C. Meyer<sup>29</sup>,
W.T. Meyer<sup>64</sup>, J. Miao<sup>32d</sup>, S. Michal<sup>29</sup>, L. Micu<sup>25a</sup>, R.P. Middleton<sup>129</sup>, P. Miele<sup>29</sup>, S. Migas<sup>73</sup>, L. Mijović<sup>41</sup>, G. Mikenberg<sup>171</sup>,
M. Mikestikova<sup>125</sup>, M. Mikuž<sup>74</sup>, D.W. Miller<sup>143</sup>, R.J. Miller<sup>88</sup>, W.J. Mills<sup>168</sup>, C. Mills<sup>57</sup>, A. Milov<sup>171</sup>, D.A. Milstead<sup>146a,146b</sup>,
D. Milstein<sup>171</sup>, A.A. Minaenko<sup>128</sup>, M. Miñano<sup>167</sup>, I.A. Minashvili<sup>65</sup>, A.I. Mincer<sup>108</sup>, B. Mindur<sup>37</sup>, M. Mineev<sup>65</sup>, Y. Ming<sup>130</sup>,

27

L.M. Mir<sup>11</sup>, G. Mirabelli<sup>132a</sup>, L. Miralles Verge<sup>11</sup>, A. Misiejuk<sup>76</sup>, J. Mitrevski<sup>137</sup>, G.Y. Mitrofanov<sup>128</sup>, V.A. Mitsou<sup>167</sup>, S. Mitsui<sup>66</sup>, P.S. Miyagawa<sup>139</sup>, K. Miyazaki<sup>67</sup>, J.U. Mjörnmark<sup>79</sup>, T. Moa<sup>146a,146b</sup>, P. Mockett<sup>138</sup>, S. Moed<sup>57</sup>, V. Moeller<sup>27</sup>, S. Mitsui<sup>66</sup>, P.S. Miyagawa<sup>139</sup>, K. Miyazaki<sup>67</sup>, J.U. Mjörnmark<sup>79</sup>, T. Moa<sup>146a,146b</sup>, P. Mockett<sup>138</sup>, S. Moed<sup>57</sup>, V. Moeller<sup>27</sup>, K. Mönig<sup>41</sup>, N. Möser<sup>20</sup>, S. Mohapatra<sup>148</sup>, W. Mohr<sup>48</sup>, S. Mohrdieck-Möck<sup>99</sup>, A.M. Moisseev<sup>128,\*</sup>, R. Moles-Valls<sup>167</sup>, J. Molina-Perez<sup>29</sup>, J. Monk<sup>77</sup>, E. Monnier<sup>83</sup>, S. Montesano<sup>89a,89b</sup>, F. Monticelli<sup>70</sup>, S. Monzani<sup>19a,19b</sup>, R.W. Moore<sup>2</sup>, G.F. Moorhead<sup>86</sup>, C. Mora Herrera<sup>49</sup>, A. Moraes<sup>53</sup>, N. Morange<sup>136</sup>, J. Morel<sup>54</sup>, G. Morello<sup>36a,36b</sup>, D. Moreno<sup>81</sup>, M. Moreno Llácer<sup>167</sup>, P. Morettini<sup>50a</sup>, M. Morii<sup>57</sup>, J. Morin<sup>75</sup>, Y. Morita<sup>66</sup>, A.K. Morley<sup>29</sup>, G. Mornacchi<sup>29</sup>, S.V. Morozov<sup>96</sup>, J.D. Morris<sup>75</sup>, L. Morvaj<sup>101</sup>, H.G. Moser<sup>99</sup>, M. Mosidze<sup>51</sup>, J. Moss<sup>109</sup>, R. Mount<sup>143</sup>, E. Mountricha<sup>136</sup>, S.V. Mouraviev<sup>94</sup>, E.J.W. Moyse<sup>84</sup>, M. Mudrinic<sup>12b</sup>, F. Mueller<sup>58a</sup>, J. Mueller<sup>123</sup>, K. Mueller<sup>20</sup>, T.A. Müller<sup>98</sup>, D. Muenstermann<sup>29</sup>, A. Muir<sup>168</sup>, Y. Munwes<sup>153</sup>, W.J. Murray<sup>129</sup>, I. Mussche<sup>105</sup>, E. Musto<sup>102a,102b</sup>, A.G. Myagkov<sup>128</sup>, M. Myska<sup>125</sup>, J. Nadal<sup>11</sup>, K. Nagai<sup>160</sup>, K. Nagano<sup>66</sup>, Y. Nagasaka<sup>60</sup>, A.M. Nairz<sup>29</sup>, Y. Nakahama<sup>29</sup>, K. Nakamura<sup>155</sup>, I. Nakano<sup>110</sup>, G. Nanava<sup>20</sup>, A. Napier<sup>161</sup>, M. Nash<sup>77,c</sup>, N.R. Nation<sup>21</sup>, T. Nattermann<sup>20</sup>, T. Naumann<sup>41</sup>, G. Navarro<sup>162</sup>, H.A. Neal<sup>87</sup>, E. Nebot<sup>80</sup>, P.Yu. Nechaeva<sup>94</sup>, A. Negri<sup>119a,119b</sup>, G. Negri<sup>29</sup>, S. Nektarijevic<sup>49</sup>, S. Nelson<sup>143</sup>, T.K. Nelson<sup>143</sup>, S. Nemecek<sup>125</sup>, P. Nemethy<sup>108</sup>, A.A. Nepomuceno<sup>23a</sup>, M. Nessi<sup>29,t</sup>, S.Y. Nesterov<sup>121</sup>, M.S. Neubauer<sup>165</sup>, A. Neusiedl<sup>81</sup>, R.M. Neves<sup>108</sup>, P. Nevski<sup>24</sup>, P.R. Newman<sup>17</sup>, V. Nguyen Thi Hong<sup>136</sup>. G. Negri<sup>29</sup>, S. Nektarijevic<sup>49</sup>, S. Nelson<sup>143</sup>, T.K. Nelson<sup>143</sup>, S. Nemecek<sup>123</sup>, P. Nemethy<sup>106</sup>, A.A. Nepomuceno<sup>2,3a</sup>, M. Nessi<sup>29,4</sup>, S.Y. Nesterov<sup>121</sup>, M.S. Neubauer<sup>165</sup>, A. Neusiedl<sup>81</sup>, R.M. Neves<sup>108</sup>, P. Nevski<sup>24</sup>, P.R. Newman<sup>17</sup>, V. Nguyen Thi Hong<sup>136</sup>, R.B. Nickerson<sup>118</sup>, R. Nicolaidou<sup>136</sup>, L. Nicolas<sup>139</sup>, B. Nicquevert<sup>29</sup>, F. Niedercorn<sup>115</sup>, J. Nielsen<sup>137</sup>, T. Niinikoski<sup>29</sup>, N. Nikiforou<sup>34</sup>, A. Nikiforov<sup>15</sup>, V. Nikolaenko<sup>128</sup>, K. Nikolaev<sup>65</sup>, I. Nikolic-Audit<sup>78</sup>, K. Nikolics<sup>49</sup>, K. Nikolopoulos<sup>24</sup>, H. Nilsen<sup>48</sup>, P. Nilsson<sup>7</sup>, Y. Ninomiya<sup>155</sup>, A. Nisati<sup>132a</sup>, T. Nishiyama<sup>67</sup>, R. Nisius<sup>99</sup>, L. Nodulman<sup>5</sup>, M. Nomachi<sup>116</sup>, I. Nomidis<sup>154</sup>, M. Nordberg<sup>29</sup>, B. Nordkvist<sup>146a,146b</sup>, P.R. Norton<sup>129</sup>, J. Novakova<sup>126</sup>, M. Nozaki<sup>66</sup>, M. Nožička<sup>41</sup>, L. Nozka<sup>113</sup>, I. Nomidis<sup>154</sup>, M. Nordberg<sup>29</sup>, B. Nordkvist<sup>146a,146b</sup>, P.R. Norton<sup>129</sup>, J. Novakova<sup>126</sup>, M. Nozaki<sup>66</sup>, M. Nožička<sup>41</sup>, L. Nozka<sup>115</sup>, I.M. Nugent<sup>159a</sup>, A.-E. Nuncio-Quiroz<sup>20</sup>, G. Nunes Hanninger<sup>86</sup>, T. Nunnemann<sup>98</sup>, E. Nurse<sup>77</sup>, T. Nyman<sup>29</sup>, B.J. O'Brien<sup>45</sup>, S.W. O'Neale<sup>17,\*</sup>, D.C. O'Neil<sup>142</sup>, V. O'Shea<sup>53</sup>, F.G. Oakham<sup>28,e</sup>, H. Oberlack<sup>99</sup>, J. Ocariz<sup>78</sup>, A. Ochi<sup>67</sup>, S. Oda<sup>155</sup>, S. Odaka<sup>66</sup>, J. Odier<sup>83</sup>, H. Ogren<sup>61</sup>, A. Oh<sup>82</sup>, S.H. Oh<sup>44</sup>, C.C. Ohm<sup>146a,146b</sup>, T. Ohshima<sup>101</sup>, H. Ohshita<sup>140</sup>, T.K. Ohska<sup>66</sup>, T. Ohsugi<sup>59</sup>, S. Okada<sup>67</sup>, H. Okawa<sup>163</sup>, Y. Okumura<sup>101</sup>, T. Okuyama<sup>155</sup>, M. Olcese<sup>50a</sup>, A.G. Olchevski<sup>65</sup>, M. Oliveira<sup>124a,h</sup>, D. Oliveira Damazio<sup>24</sup>, E. Oliver Garcia<sup>167</sup>, D. Olivito<sup>120</sup>, A. Olszewski<sup>38</sup>, J. Olszowska<sup>38</sup>, C. Omachi<sup>67</sup>, A. Onofre<sup>124a,u</sup>, P.U.E. Onyisi<sup>30</sup>, C.J. Oram<sup>159a</sup>, M.J. Oreglia<sup>30</sup>, Y. Oren<sup>153</sup>, D. Orestano<sup>134a,134b</sup>, I. Orlov<sup>107</sup>, C. Oropeza Barrera<sup>53</sup>, R.S. Orr<sup>158</sup>, B. Osculati<sup>50a,50b</sup>, R. Ospanov<sup>120</sup>, C. Osuna<sup>11</sup>, G. Otero y Garzon<sup>26</sup>, J.P Ottersbach<sup>105</sup>, M. Ouchrif<sup>135d</sup>, F. Ould-Saada<sup>117</sup>, D. Divera Damazio<sup>+</sup>, E. Oliver Garcia<sup>-, D</sup>, Olivito<sup>+, T</sup>, A. Olszewska<sup>+, J</sup>, Oliszowska<sup>+, J</sup>, Olszowska<sup>+, J</sup>, C. Omach<sup>+, J</sup>, A. Onore<sup>+, L, J</sup>, S. Ort<sup>158</sup>, P. Cuel, Sandov<sup>120</sup>, C. Osuna<sup>11</sup>, G. Otero y Garzon<sup>26</sup>, J.P Ottersbach<sup>105</sup>, M. Ouchrif<sup>153</sup>, F. Oudd-Saada<sup>117</sup>, A. Ouraro<sup>136</sup>, D. Ostanov<sup>120</sup>, C. Osuna<sup>11</sup>, G. Otero y Garzon<sup>26</sup>, J.P Ottersbach<sup>105</sup>, M. Ouchrif<sup>153</sup>, F. Oudd-Saada<sup>117</sup>, A. Ouraro<sup>137</sup>, D. Palme<sup>238</sup>, S. Owen<sup>139</sup>, V.E. Ozcan<sup>184</sup>, N. Ozturk<sup>-7</sup>, A. Pacheco Pages<sup>11</sup>, C. Padila Aranda<sup>11</sup>, S. Pagan Griso<sup>14</sup>, E. Paganis<sup>139</sup>, F. Paige<sup>24</sup>, K. Pajchel<sup>117</sup>, G. Palacino<sup>1596</sup>, C.P. Paleari<sup>6</sup>, S. Palestini<sup>97</sup>, D. Pante<sup>258</sup>, M. Pantes<sup>258</sup>, M. Panusk<sup>500</sup>, J.D. Palmet<sup>17</sup>, Y.B. Pan<sup>172</sup>, E. Panagiotopoulov<sup>9</sup>, B. Paras<sup>1134</sup>, N. Pantk<sup>517</sup>, M.A. Parter<sup>27</sup>, F. Parodi<sup>30a,260</sup>, J.A. Parsons<sup>34</sup>, U. Parzefall<sup>48</sup>, E. Pasqualucci<sup>132a</sup>, A. Passer<sup>1144</sup>, F. Pastor<sup>2144</sup>, M.I. Pedraza Morales<sup>172</sup>, S. V. Peleganchuk<sup>107</sup>, H. Peng<sup>228</sup>, R. Pengo<sup>29</sup>, A. Penson<sup>34</sup>, J. Penvell<sup>61</sup>, M. Perantoni<sup>23a</sup>, K. Percz<sup>34,4</sup>, T. Percz Cavalcanti<sup>41</sup>, E. Perez Codina<sup>14</sup>, M.T. Perez García-Estal<sup>167</sup>, V. Perz Reale<sup>34</sup>, J. Penvell<sup>61</sup>, M. Perantoni<sup>23a</sup>, K. Percz<sup>34,4</sup>, T. Percz Cavalcanti<sup>41</sup>, E. Perce<sup>34,4</sup>, M.J. Pedraza Morales<sup>172</sup>, S. Perciolo<sup>152a</sup>, F. Petrucko<sup>154a</sup>, 149, D. Petschul<sup>11</sup>, J. Petresn<sup>27</sup>, T. C. Petresn<sup>35</sup>, E. Petil<sup>35</sup>, A. Petrido<sup>154</sup>, C. Putrido<sup>154</sup>, C. Putrido<sup>154</sup>, D. Petschul<sup>14</sup>, M. Petera<sup>172</sup>, K. Penzo<sup>35</sup>, E. Petrido<sup>155</sup>, P. Devtido<sup>154</sup>, C. Putrido<sup>154</sup>, P. Petrodo<sup>152a</sup>, F. Petrucko<sup>154a</sup>, P. Neider<sup>118</sup>, M. Pinamonti<sup>164a,164c</sup>, A. Pinder<sup>118</sup>, J. L. Pinfol<sup>24</sup>, J. N. Plessia<sup>175</sup>, J. P. Nuthilips<sup>27</sup>, P. W. Phillips<sup>129</sup>, G. Piacatyte<sup>14</sup>, M. Pinamonti<sup>164a,164c</sup>, A. Pinder<sup>118</sup>, J. Pinf<sup>25</sup>, G. Polsel<sup>107</sup>, I. Nuthi<sup>155</sup>, P. Podtysk<sup>33</sup>, J. Polif<sup>154</sup>, P. Portug<sup>47</sup>, G. Cosch<sup>216</sup>, G. C. Poseplo<sup>175</sup>, J. Polito<sup>155</sup>, P. Podtysk<sup>33</sup>, J. Polito<sup>157</sup>, T. Pottos<sup>179</sup>, P. Pinz<sup>156</sup>, D. Pomeroy<sup>22</sup>, K. Pomine<sup>59</sup>, L. Pontec<sup>154</sup>, A. Polpere<sup>156</sup>, A.

N.A. Rusakovich<sup>65</sup>, D.R. Rust<sup>61</sup>, J.P. Rutherfoord<sup>6</sup>, C. Ruwiedel<sup>14</sup>, P. Ruzicka<sup>125</sup>, Y.F. Ryabov<sup>121</sup>, V. Ryadovikov<sup>128</sup>, P. Ryan<sup>88</sup>, M. Rybar<sup>126</sup>, G. Rybkin<sup>115</sup>, N.C. Ryder<sup>118</sup>, S. Rzaeva<sup>10</sup>, A.F. Saavedra<sup>150</sup>, I. Sadeh<sup>153</sup>, H.F-W. Sadrozinski<sup>137</sup>, R. Sadykov<sup>65</sup>, F. Safai Tehrani<sup>132a,132b</sup>, H. Sakamoto<sup>155</sup>, G. Salamanna<sup>75</sup>, A. Salamon<sup>133a</sup>, M. Saleem<sup>111</sup>, D. Salihagic<sup>99</sup>, A. Salnikov<sup>143</sup>, J. Salt<sup>167</sup>, B.M. Salvachua Ferrando<sup>5</sup>, D. Salvatore<sup>36a,36b</sup>, F. Salvatore<sup>149</sup>, A. Salvucci<sup>104</sup>, A. Salzburger<sup>29</sup>, D. Sampsonidis<sup>154</sup>, B.H. Samset<sup>117</sup>, A. Sanchez<sup>102a,102b</sup>, H. Sandaker<sup>13</sup>, H.G. Sander<sup>81</sup>, M.P. Sanders<sup>98</sup>, M. Sandhoff<sup>174</sup>, T. Sandoval<sup>27</sup>, C. Sandoval<sup>162</sup>, R. Sandstroem<sup>99</sup>, S. Sandvoss<sup>174</sup>, D.P.C. Sankey<sup>129</sup>, A. Sansoni<sup>47</sup>, C. Santamarina Rios<sup>85</sup>, C. Santoni<sup>33</sup>, R. Santonico<sup>133a,133b</sup>, H. Santos<sup>124a</sup>, J.G. Saraiva<sup>124a,b</sup>, T. Sarangi<sup>172</sup>, E. Sarkisyan-Grinbaum<sup>7</sup>, F. Sarri<sup>122a,122b</sup>, G. Sartisohn<sup>174</sup>, O. Sasaki<sup>66</sup>, T. Sasaki<sup>66</sup>, N. Sasao<sup>68</sup>, I. Satsounkevitch<sup>90</sup>, G. Sauvage<sup>4</sup>, E. Sauvan<sup>4</sup>, J.B. Sauvan<sup>115</sup>, P. Savard<sup>158,e</sup>, V. Savinov<sup>123</sup>, D.O. Savu<sup>29</sup>, P. Savva<sup>9</sup>, L. Sawyer<sup>24,l</sup>, D.H. Saxon<sup>53</sup>, L.P. Says<sup>33</sup>, C. Sbarra<sup>19a,19b</sup>, A. Sbrizzi<sup>19a,19b</sup>, O. Scallon<sup>93</sup>, D.A. Scannicchio<sup>163</sup>, J. Schaarschmidt<sup>115</sup>, P. Schacht<sup>99</sup>, U. Schäfer<sup>81</sup>, S. Schaepe<sup>20</sup>, S. Schaetzel<sup>58b</sup>, A.C. Schaffer<sup>115</sup>, D. Schaile<sup>98</sup>, R.D. Schamberger<sup>148</sup>, A.G. Schamov<sup>107</sup>, V. Scharf<sup>58a</sup>, V.A. Schegelsky<sup>121</sup>, D. Scheirich<sup>87</sup>, M. Scherrau<sup>163</sup>, M.I. Scherzer<sup>14</sup>, C. Schiayi<sup>50a,50b</sup>, J. Schieck<sup>98</sup>, M. Schioppa<sup>36a,36b</sup>, S. Schlenker<sup>29</sup>, J.L. Scherzet<sup>15</sup>, Sonaberger<sup>148</sup>, A.G. Schamov<sup>107</sup>, V. Scharf<sup>58a</sup>, V.A. Schegelsky<sup>121</sup>, D. Scheirich<sup>87</sup>, M. Scherrau<sup>163</sup>, M.I. Scherzer<sup>14</sup>, C. Schiayi<sup>50a,50b</sup>, J. Schiack<sup>98</sup>, M. Schioppa<sup>36a,36b</sup>, S. Schlenker<sup>29</sup>, J.L. Schereth<sup>5</sup>, O. Scallon<sup>11</sup>, J. A. Scannicho<sup>11</sup>, J. Schaarberger<sup>11</sup>, S. C. Schaffer<sup>11</sup>, S. C. Schaffer<sup>11</sup>, J. Schaffer<sup>11</sup> M. Schernau<sup>163</sup>, M.I. Scherzer<sup>14</sup>, C. Schiavi<sup>50a,50b</sup>, J. Schieck<sup>98</sup>, M. Schioppa<sup>36a,36b</sup>, S. Schlenker<sup>29</sup>, J.L. Schlereth<sup>5</sup>,

29
G. Unel<sup>163</sup>, Y. Unno<sup>65</sup>, D. Urbaniec<sup>34</sup>, E. Urkovsky<sup>153</sup>, P. Urrejola<sup>31</sup>, G. Usai<sup>7</sup>, M. Uslenghi<sup>1194,1195</sup>, I. Vacavant<sup>83</sup>, Vacek<sup>127</sup>, B. Vachon<sup>85</sup>, S. Valker<sup>147</sup>, I. Valenta<sup>128</sup>, P. Valente<sup>122a</sup>, S. Valentinetti<sup>194,1195</sup>, S. Valkar<sup>12</sup>, F. Valladolid Gallego<sup>167</sup>, S. Vallecorsa<sup>152</sup>, J.A. Valls Ferrer<sup>167</sup>, H. van der Graaf<sup>105</sup>, E. van der Krauij<sup>106</sup>, R. Van Der Leuw<sup>165</sup>, E. Varallell<sup>87</sup>, O. Vandoni<sup>29</sup>, A. Vaniachine<sup>5</sup>, P. Vankov<sup>41</sup>, F. Van Gemmerren<sup>5</sup>, Z. van Kestern<sup>167</sup>, I. van Vulere<sup>168</sup>, W. Vandell<sup>87</sup>, G. Vandoni<sup>29</sup>, A. Vaniachine<sup>5</sup>, P. Vankov<sup>41</sup>, F. Vannecci<sup>28</sup>, F. Varella Rodriguez<sup>29</sup>, R. Vari<sup>123</sup>, E.W. Varnes<sup>6</sup>, D. Varoucha<sup>14</sup>, Varapet<sup>177</sup>, K. L. Yarvel<sup>150</sup>, V. Vances<sup>178</sup>, D. Venturi<sup>168</sup>, M. Venturi<sup>169</sup>, V. Varnes<sup>61</sup>, Veloso<sup>134</sup>, Varapet<sup>178</sup>, M. Verturi<sup>167</sup>, J. Vatil<sup>161</sup>, C. Veltishi<sup>177</sup>, M. Viti<sup>21</sup>, I. Virchu<sup>173</sup>, G. Vegni<sup>38</sup>, M. Venturi<sup>168</sup>, V. Veresci<sup>1198</sup>, M. Vertucci<sup>138</sup>, Veterke<sup>161</sup>, J. Virchu<sup>164</sup>, O. Volp<sup>156</sup>, G. Volpi<sup>167</sup>, M. Voltishi<sup>178</sup>, M. Venturi<sup>168</sup>, V. Veresci<sup>1199</sup>, M. Verducci<sup>138</sup>, Veloso<sup>134</sup>, V. Voreke<sup>174</sup>, O. Volrell<sup>174</sup>, A. Venturi<sup>274,79</sup>, D. Venturi<sup>148</sup>, F. Vives Vague<sup>11</sup>, S. Vakobo<sup>5</sup>, M. Vassebeld<sup>73</sup>, N. Varay<sup>264</sup>, V. Vorobel<sup>126</sup>, A. P. Vorobel<sup>126</sup>, A. P. Vorobel<sup>126</sup>, A. Vorreswijt<sup>166</sup>, T. Vu Anh<sup>81</sup>, R. Vuillermet<sup>27</sup>, N. Vasio<sup>276</sup>, Varajes<sup>178</sup>, J. Wang<sup>250</sup>, J. Vange<sup>179</sup>, J. Vange<sup>184</sup>, V. Vatoso<sup>175</sup>, M. Varay<sup>175</sup>, M. Vereswijt<sup>166</sup>, T. Vu Anh<sup>81</sup>, R. Vuillermet<sup>27</sup>, P. Wagne<sup>174</sup>, O. Valle<sup>174</sup>, V. Walle<sup>174</sup>, M. Wang<sup>172</sup>, M. Warshi<sup>176</sup>, T. Vu Anh<sup>81</sup>, Y. Vuille<sup>174</sup>, P. Wagne<sup>175</sup>, A. Wangh<sup>175</sup>, M. Vereswijt<sup>166</sup>, T. Vu Anh<sup>81</sup>, K. Vuille<sup>174</sup>, R. Walke<sup>78</sup>, Walke<sup>78</sup>, J. Walde<sup>71</sup>, R. Walke<sup>78</sup>, Walke<sup>78</sup>, J. Walde<sup>71</sup>, R. Walke<sup>78</sup>, W. Wartovi<sup>7</sup>, T. Wang<sup>150</sup>, J. Vang<sup>150</sup>, J. Vang<sup>150</sup>, J. Vang<sup>150</sup>, J. Walke<sup>78</sup>, M. Wenturi<sup>85</sup>, M. Warth<sup>175</sup>, T. Vu Anh<sup>81</sup>, K. Walki<sup>175</sup>, J. Vang<sup>150</sup>, J. Vang<sup>150</sup>, J. Wang<sup>150</sup>, J. Wang<sup>150</sup>, J. Wang<sup>150</sup>, J. Walke<sup>78</sup>, J. Walke<sup>78</sup>, J. Walke<sup>78</sup>,

<sup>1</sup> University at Albany, Albany NY, United States of America

<sup>2</sup> Department of Physics, University of Alberta, Edmonton AB, Canada

<sup>3</sup> <sup>(a)</sup>Department of Physics, Ankara University, Ankara; <sup>(b)</sup>Department of Physics, Dumlupinar University, Kutahya;

<sup>(c)</sup>Department of Physics, Gazi University, Ankara; <sup>(d)</sup>Division of Physics, TOBB University of Economics and Technology, Ankara; <sup>(e)</sup>Turkish Atomic Energy Authority, Ankara, Turkey

- <sup>4</sup> LAPP, CNRS/IN2P3 and Université de Savoie, Annecy-le-Vieux, France
- <sup>5</sup> High Energy Physics Division, Argonne National Laboratory, Argonne IL, United States of America
- <sup>6</sup> Department of Physics, University of Arizona, Tucson AZ, United States of America
- <sup>7</sup> Department of Physics, The University of Texas at Arlington, Arlington TX, United States of America
- <sup>8</sup> Physics Department, University of Athens, Athens, Greece
- <sup>9</sup> Physics Department, National Technical University of Athens, Zografou, Greece
- <sup>10</sup> Institute of Physics, Azerbaijan Academy of Sciences, Baku, Azerbaijan
- <sup>11</sup> Institut de Física d'Altes Energies and Universitat Autònoma de Barcelona and ICREA, Barcelona, Spain
- <sup>12</sup> (a) Institute of Physics, University of Belgrade, Belgrade; <sup>(b)</sup> Vinca Institute of Nuclear Sciences, Belgrade, Serbia
- <sup>13</sup> Department for Physics and Technology, University of Bergen, Bergen, Norway
- <sup>14</sup> Physics Division, Lawrence Berkeley National Laboratory and University of California, Berkeley CA, United States of America
- <sup>15</sup> Department of Physics, Humboldt University, Berlin, Germany
- <sup>16</sup> Albert Einstein Center for Fundamental Physics and Laboratory for High Energy Physics, University of Bern, Bern, Switzerland
- <sup>17</sup> School of Physics and Astronomy, University of Birmingham, Birmingham, United Kingdom

<sup>18</sup> (a) Department of Physics, Bogazici University, Istanbul; <sup>(b)</sup> Division of Physics, Dogus University, Istanbul; <sup>(c)</sup> Department of Physics Engineering, Gaziantep University, Gaziantep; <sup>(d)</sup>Department of Physics, Istanbul Technical University, Istanbul, Turkey

- <sup>19</sup> (*a*) INFN Sezione di Bologna; <sup>(b)</sup> Dipartimento di Fisica, Università di Bologna, Bologna, Italy
- <sup>20</sup> Physikalisches Institut, University of Bonn, Bonn, Germany
- <sup>21</sup> Department of Physics, Boston University, Boston MA, United States of America
- <sup>22</sup> Department of Physics, Brandeis University, Waltham MA, United States of America
- <sup>23</sup> (a) Universidade Federal do Rio De Janeiro COPPE/EE/IF, Rio de Janeiro; <sup>(b)</sup>Federal University of Juiz de Fora (UFJF), Juiz de Fora; <sup>(c)</sup>Federal University of Sao Joao del Rei (UFSJ), Sao Joao del Rei; <sup>(d)</sup>Instituto de Fisica, Universidade de Sao Paulo. Sao Paulo, Brazil
- <sup>24</sup> Physics Department, Brookhaven National Laboratory, Upton NY, United States of America

<sup>25</sup> (a) National Institute of Physics and Nuclear Engineering, Bucharest; <sup>(b)</sup>University Politehnica Bucharest, Bucharest; <sup>(c)</sup>West University in Timisoara, Timisoara, Romania

- <sup>26</sup> Departamento de Física, Universidad de Buenos Aires, Buenos Aires, Argentina
- <sup>27</sup> Cavendish Laboratory, University of Cambridge, Cambridge, United Kingdom
- <sup>28</sup> Department of Physics, Carleton University, Ottawa ON, Canada
- <sup>29</sup> CERN, Geneva, Switzerland
- <sup>30</sup> Enrico Fermi Institute, University of Chicago, Chicago IL, United States of America

<sup>31</sup> (a) Departamento de Fisica, Pontificia Universidad Católica de Chile, Santiago; <sup>(b)</sup> Departamento de Física, Universidad Técnica Federico Santa María, Valparaíso, Chile

<sup>32</sup> (a) Institute of High Energy Physics, Chinese Academy of Sciences, Beijing; <sup>(b)</sup>Department of Modern Physics, University of Science and Technology of China, Anhui; <sup>(c)</sup>Department of Physics, Nanjing University, Jiangsu; <sup>(d)</sup>High Energy Physics Group, Shandong University, Shandong, China

<sup>33</sup> Laboratoire de Physique Corpusculaire, Clermont Université and Université Blaise Pascal and CNRS/IN2P3, Aubiere Cedex, France

- <sup>34</sup> Nevis Laboratory, Columbia University, Irvington NY, United States of America
- <sup>35</sup> Niels Bohr Institute, University of Copenhagen, Kobenhavn, Denmark
- <sup>36</sup> (a) INFN Gruppo Collegato di Cosenza; <sup>(b)</sup> Dipartimento di Fisica, Università della Calabria, Arcavata di Rende, Italy
   <sup>37</sup> Faculty of Physics and Applied Computer Science, AGH-University of Science and Technology, Krakow, Poland
- <sup>38</sup> The Henryk Niewodniczanski Institute of Nuclear Physics, Polish Academy of Sciences, Krakow, Poland
- <sup>39</sup> Physics Department, Southern Methodist University, Dallas TX, United States of America
- <sup>40</sup> Physics Department, University of Texas at Dallas, Richardson TX, United States of America
- <sup>41</sup> DESY, Hamburg and Zeuthen, Germany
- <sup>42</sup> Institut für Experimentelle Physik IV, Technische Universität Dortmund, Dortmund, Germany
- <sup>43</sup> Institut für Kern- und Teilchenphysik, Technical University Dresden, Dresden, Germany
- <sup>44</sup> Department of Physics, Duke University, Durham NC, United States of America
- <sup>45</sup> SUPA School of Physics and Astronomy, University of Edinburgh, Edinburgh, United Kingdom
- <sup>46</sup> Fachhochschule Wiener Neustadt, Johannes Gutenbergstrasse 3, 2700 Wiener Neustadt, Austria
- <sup>47</sup> INFN Laboratori Nazionali di Frascati, Frascati, Italy
- <sup>48</sup> Fakultät für Mathematik und Physik, Albert-Ludwigs-Universität, Freiburg i.Br., Germany 49. Fakultät für Mathematik und Physik, Albert-Ludwigs-Universität, Freiburg i.Br., Germany
- <sup>49</sup> Section de Physique, Université de Genève, Geneva, Switzerland
- <sup>50</sup> (a) INFN Sezione di Genova; <sup>(b)</sup> Dipartimento di Fisica, Università di Genova, Genova, Italy
- <sup>51</sup> Institute of Physics and HEP Institute, Georgian Academy of Sciences and Tbilisi State University, Tbilisi, Georgia
- <sup>52</sup> II Physikalisches Institut, Justus-Liebig-Universität Giessen, Giessen, Germany
- <sup>53</sup> SUPA School of Physics and Astronomy, University of Glasgow, Glasgow, United Kingdom
- <sup>54</sup> II Physikalisches Institut, Georg-August-Universität, Göttingen, Germany

<sup>55</sup> Laboratoire de Physique Subatomique et de Cosmologie, Université Joseph Fourier and CNRS/IN2P3 and Institut National Polytechnique de Grenoble, Grenoble, France

- <sup>56</sup> Department of Physics, Hampton University, Hampton VA, United States of America
- <sup>57</sup> Laboratory for Particle Physics and Cosmology, Harvard University, Cambridge MA, United States of America

<sup>58</sup> (*a*)Kirchhoff-Institut für Physik, Ruprecht-Karls-Universität Heidelberg, Heidelberg; <sup>(b)</sup>Physikalisches Institut,

#### Ruprecht-Karls-Universität Heidelberg, Heidelberg; (c)ZITI Institut für technische Informatik, Ruprecht-Karls-Universität Heidelberg, Mannheim, Germany

- <sup>59</sup> Faculty of Science, Hiroshima University, Hiroshima, Japan
- <sup>60</sup> Faculty of Applied Information Science, Hiroshima Institute of Technology, Hiroshima, Japan
- <sup>61</sup> Department of Physics, Indiana University, Bloomington IN, United States of America
- <sup>62</sup> Institut für Astro- und Teilchenphysik, Leopold-Franzens-Universität, Innsbruck, Austria
- <sup>63</sup> University of Iowa, Iowa City IA, United States of America

- <sup>64</sup> Department of Physics and Astronomy, Iowa State University, Ames IA, United States of America
- <sup>65</sup> Joint Institute for Nuclear Research, JINR Dubna, Dubna, Russia
- <sup>66</sup> KEK, High Energy Accelerator Research Organization, Tsukuba, Japan
- <sup>67</sup> Graduate School of Science, Kobe University, Kobe, Japan
- <sup>68</sup> Faculty of Science, Kyoto University, Kyoto, Japan
- <sup>69</sup> Kyoto University of Education, Kyoto, Japan
- <sup>70</sup> Instituto de Física La Plata, Universidad Nacional de La Plata and CONICET, La Plata, Argentina
- <sup>71</sup> Physics Department, Lancaster University, Lancaster, United Kingdom
- <sup>72</sup> (a) INFN Sezione di Lecce; <sup>(b)</sup> Dipartimento di Fisica, Università del Salento, Lecce, Italy
- <sup>73</sup> Oliver Lodge Laboratory, University of Liverpool, Liverpool, United Kingdom
- <sup>74</sup> Department of Physics, Jožef Stefan Institute and University of Ljubljana, Ljubljana, Slovenia
- <sup>75</sup> Department of Physics, Queen Mary University of London, London, United Kingdom
- <sup>76</sup> Department of Physics, Royal Holloway University of London, Surrey, United Kingdom
- <sup>77</sup> Department of Physics and Astronomy, University College London, London, United Kingdom
- <sup>78</sup> Laboratoire de Physique Nucléaire et de Hautes Energies, UPMC and Université Paris-Diderot and CNRS/IN2P3, Paris,

France

- <sup>79</sup> Fysiska institutionen, Lunds universitet, Lund, Sweden
- <sup>80</sup> Departamento de Fisica Teorica C-15, Universidad Autonoma de Madrid, Madrid, Spain
- <sup>81</sup> Institut für Physik, Universität Mainz, Mainz, Germany
- <sup>82</sup> School of Physics and Astronomy, University of Manchester, Manchester, United Kingdom
- <sup>83</sup> CPPM, Aix-Marseille Université and CNRS/IN2P3, Marseille, France
- <sup>84</sup> Department of Physics, University of Massachusetts, Amherst MA, United States of America
- <sup>85</sup> Department of Physics, McGill University, Montreal QC, Canada
- <sup>86</sup> School of Physics, University of Melbourne, Victoria, Australia
- <sup>87</sup> Department of Physics, The University of Michigan, Ann Arbor MI, United States of America
- <sup>88</sup> Department of Physics and Astronomy, Michigan State University, East Lansing MI, United States of America
- <sup>89</sup> (a) INFN Sezione di Milano; <sup>(b)</sup> Dipartimento di Fisica, Università di Milano, Milano, Italy
- <sup>90</sup> B.I. Stepanov Institute of Physics, National Academy of Sciences of Belarus, Minsk, Republic of Belarus
- <sup>91</sup> National Scientific and Educational Centre for Particle and High Energy Physics, Minsk, Republic of Belarus
- <sup>92</sup> Department of Physics, Massachusetts Institute of Technology, Cambridge MA, United States of America
- <sup>93</sup> Group of Particle Physics, University of Montreal, Montreal QC, Canada
- <sup>94</sup> P.N. Lebedev Institute of Physics, Academy of Sciences, Moscow, Russia
- <sup>95</sup> Institute for Theoretical and Experimental Physics (ITEP), Moscow, Russia
- <sup>96</sup> Moscow Engineering and Physics Institute (MEPhI), Moscow, Russia
- <sup>97</sup> Skobeltsyn Institute of Nuclear Physics, Lomonosov Moscow State University, Moscow, Russia
- <sup>98</sup> Fakultät für Physik, Ludwig-Maximilians-Universität München, München, Germany
- <sup>99</sup> Max-Planck-Institut für Physik (Werner-Heisenberg-Institut), München, Germany
- <sup>100</sup> Nagasaki Institute of Applied Science, Nagasaki, Japan
- <sup>101</sup> Graduate School of Science, Nagoya University, Nagoya, Japan
- <sup>102</sup> (a) INFN Sezione di Napoli; <sup>(b)</sup> Dipartimento di Scienze Fisiche, Università di Napoli, Napoli, Italy
- <sup>103</sup> Department of Physics and Astronomy, University of New Mexico, Albuquerque NM, United States of America
- <sup>104</sup> Institute for Mathematics, Astrophysics and Particle Physics, Radboud University Nijmegen/Nikhef, Nijmegen, Netherlands
- <sup>105</sup> Nikhef National Institute for Subatomic Physics and University of Amsterdam, Amsterdam, Netherlands
- <sup>106</sup> Department of Physics, Northern Illinois University, DeKalb IL, United States of America
- <sup>107</sup> Budker Institute of Nuclear Physics (BINP), Novosibirsk, Russia
- <sup>108</sup> Department of Physics, New York University, New York NY, United States of America
- <sup>109</sup> Ohio State University, Columbus OH, United States of America
- <sup>110</sup> Faculty of Science, Okayama University, Okayama, Japan
- <sup>111</sup> Homer L. Dodge Department of Physics and Astronomy, University of Oklahoma, Norman OK, United States of America
- <sup>112</sup> Department of Physics, Oklahoma State University, Stillwater OK, United States of America
- <sup>113</sup> Palacký University, RCPTM, Olomouc, Czech Republic
- <sup>114</sup> Center for High Energy Physics, University of Oregon, Eugene OR, United States of America
- <sup>115</sup> LAL, Univ. Paris-Sud and CNRS/IN2P3, Orsay, France
- <sup>116</sup> Graduate School of Science, Osaka University, Osaka, Japan
- <sup>117</sup> Department of Physics, University of Oslo, Oslo, Norway
- <sup>118</sup> Department of Physics, Oxford University, Oxford, United Kingdom
- <sup>119</sup> <sup>(a)</sup> INFN Sezione di Pavia; <sup>(b)</sup> Dipartimento di Fisica Nucleare e Teorica, Università di Pavia, Pavia, Italy
   <sup>120</sup> Department of Physics, University of Pennsylvania, Philadelphia PA, United States of America
- <sup>121</sup> Petersburg Nuclear Physics Institute, Gatchina, Russia

<sup>122</sup> <sup>(a)</sup>INFN Sezione di Pisa; <sup>(b)</sup>Dipartimento di Fisica E. Fermi, Università di Pisa, Pisa, Italy

<sup>123</sup> Department of Physics and Astronomy, University of Pittsburgh, Pittsburgh PA, United States of America

<sup>124</sup> <sup>(a)</sup>Laboratorio de Instrumentação e Fisica Experimental de Particulas - LIP, Lisboa, Portugal; <sup>(b)</sup>Departamento de Fisica Teorica y del Cosmos and CAFPE, Universidad de Granada, Granada, Spain

<sup>125</sup> Institute of Physics, Academy of Sciences of the Czech Republic, Praha, Czech Republic

- <sup>126</sup> Faculty of Mathematics and Physics, Charles University in Prague, Praha, Czech Republic
- <sup>127</sup> Czech Technical University in Prague, Praha, Czech Republic
- <sup>128</sup> State Research Center Institute for High Energy Physics, Protvino, Russia
- <sup>129</sup> Particle Physics Department, Rutherford Appleton Laboratory, Didcot, United Kingdom
- <sup>130</sup> Physics Department, University of Regina, Regina SK, Canada
- <sup>131</sup> Ritsumeikan University, Kusatsu, Shiga, Japan

<sup>132</sup> <sup>(a)</sup>INFN Sezione di Roma I; <sup>(b)</sup>Dipartimento di Fisica, Università La Sapienza, Roma, Italy

<sup>133</sup> <sup>(a)</sup>INFN Sezione di Roma Tor Vergata; <sup>(b)</sup>Dipartimento di Fisica, Università di Roma Tor Vergata, Roma, Italy

<sup>134</sup> <sup>(a)</sup>INFN Sezione di Roma Tre; <sup>(b)</sup>Dipartimento di Fisica, Università Roma Tre, Roma, Italy

<sup>135</sup> (*a*) Faculté des Sciences Ain Chock, Réseau Universitaire de Physique des Hautes Energies - Université Hassan II,

Casablanca; <sup>(b)</sup>Centre National de l'Energie des Sciences Techniques Nucleaires, Rabat; <sup>(c)</sup>Université Cadi Ayyad, Faculté des

sciences Semlalia Département de Physique, B.P. 2390 Marrakech 40000; <sup>(d)</sup>Faculté des Sciences, Université Mohamed

Premier and LPTPM, Oujda; <sup>(e)</sup>Faculté des Sciences, Université Mohammed V, Rabat, Morocco

<sup>136</sup> DSM/IRFU (Institut de Recherches sur les Lois Fondamentales de l'Univers), CEA Saclay (Commissariat a l'Energie Atomique), Gif-sur-Yvette, France

- <sup>137</sup> Santa Cruz Institute for Particle Physics, University of California Santa Cruz, Santa Cruz CA, United States of America
- <sup>138</sup> Department of Physics, University of Washington, Seattle WA, United States of America
- <sup>139</sup> Department of Physics and Astronomy, University of Sheffield, Sheffield, United Kingdom

<sup>140</sup> Department of Physics, Shinshu University, Nagano, Japan

- <sup>141</sup> Fachbereich Physik, Universität Siegen, Siegen, Germany
- <sup>142</sup> Department of Physics, Simon Fraser University, Burnaby BC, Canada

<sup>143</sup> SLAC National Accelerator Laboratory, Stanford CA, United States of America

<sup>144</sup> <sup>(a)</sup>Faculty of Mathematics, Physics & Informatics, Comenius University, Bratislava; <sup>(b)</sup>Department of Subnuclear Physics,

Institute of Experimental Physics of the Slovak Academy of Sciences, Kosice, Slovak Republic <sup>145</sup> <sup>(a)</sup>Department of Physics, University of Johannesburg, Johannesburg; <sup>(b)</sup>School of Physics, University of the Witwatersrand, Johannesburg, South Africa

<sup>146</sup> <sup>(a)</sup>Department of Physics, Stockholm University; <sup>(b)</sup>The Oskar Klein Centre, Stockholm, Sweden

<sup>147</sup> Physics Department, Royal Institute of Technology, Stockholm, Sweden

<sup>148</sup> Department of Physics and Astronomy, Stony Brook University, Stony Brook NY, United States of America

<sup>149</sup> Department of Physics and Astronomy, University of Sussex, Brighton, United Kingdom

- <sup>150</sup> School of Physics, University of Sydney, Sydney, Australia
- <sup>151</sup> Institute of Physics, Academia Sinica, Taipei, Taiwan
- <sup>152</sup> Department of Physics, Technion: Israel Inst. of Technology, Haifa, Israel
- <sup>153</sup> Raymond and Beverly Sackler School of Physics and Astronomy, Tel Aviv University, Tel Aviv, Israel
- <sup>154</sup> Department of Physics, Aristotle University of Thessaloniki, Thessaloniki, Greece
- <sup>155</sup> International Center for Elementary Particle Physics and Department of Physics. The University of Tokyo, Tokyo, Japan
- <sup>156</sup> Graduate School of Science and Technology, Tokyo Metropolitan University, Tokyo, Japan

<sup>157</sup> Department of Physics, Tokyo Institute of Technology, Tokyo, Japan

<sup>158</sup> Department of Physics, University of Toronto, Toronto ON, Canada

<sup>159</sup> (a) TRIUMF, Vancouver BC; <sup>(b)</sup> Department of Physics and Astronomy, York University, Toronto ON, Canada

<sup>160</sup> Institute of Pure and Applied Sciences, University of Tsukuba, Ibaraki, Japan

<sup>161</sup> Science and Technology Center, Tufts University, Medford MA, United States of America

<sup>162</sup> Centro de Investigaciones, Universidad Antonio Narino, Bogota, Colombia

<sup>163</sup> Department of Physics and Astronomy, University of California Irvine, Irvine CA, United States of America

<sup>164</sup> (a) INFN Gruppo Collegato di Udine; <sup>(b)</sup>ICTP, Trieste; <sup>(c)</sup>Dipartimento di Fisica, Università di Udine, Udine, Italy

<sup>165</sup> Department of Physics, University of Illinois, Urbana IL, United States of America

<sup>166</sup> Department of Physics and Astronomy, University of Uppsala, Uppsala, Sweden

<sup>167</sup> Instituto de Física Corpuscular (IFIC) and Departamento de Física Atómica, Molecular y Nuclear and Departamento de

Ingenierá Electrónica and Instituto de Microelectrónica de Barcelona (IMB-CNM), University of Valencia and CSIC, Valencia, Spain

- <sup>168</sup> Department of Physics, University of British Columbia, Vancouver BC, Canada
- <sup>169</sup> Department of Physics and Astronomy, University of Victoria, Victoria BC, Canada
- <sup>170</sup> Waseda University, Tokyo, Japan

The ATLAS Collaboration: Performance of Missing Transverse Momentum Reconstruction at  $\sqrt{s} = 7$  TeV

- <sup>171</sup> Department of Particle Physics, The Weizmann Institute of Science, Rehovot, Israel
- <sup>172</sup> Department of Physics, University of Wisconsin, Madison WI, United States of America
- <sup>173</sup> Fakultät für Physik und Astronomie, Julius-Maximilians-Universität, Würzburg, Germany
- <sup>174</sup> Fachbereich C Physik, Bergische Universität Wuppertal, Wuppertal, Germany
- <sup>175</sup> Department of Physics, Yale University, New Haven CT, United States of America
- <sup>176</sup> Yerevan Physics Institute, Yerevan, Armenia
- <sup>177</sup> Domaine scientifique de la Doua, Centre de Calcul CNRS/IN2P3, Villeurbanne Cedex, France
- <sup>a</sup> Also at Laboratorio de Instrumentacao e Fisica Experimental de Particulas LIP, Lisboa, Portugal
- <sup>b</sup> Also at Faculdade de Ciencias and CFNUL, Universidade de Lisboa, Lisboa, Portugal
- <sup>c</sup> Also at Particle Physics Department, Rutherford Appleton Laboratory, Didcot, United Kingdom
- <sup>d</sup> Also at CPPM, Aix-Marseille Université and CNRS/IN2P3, Marseille, France
- <sup>e</sup> Also at TRIUMF, Vancouver BC, Canada
- <sup>f</sup> Also at Department of Physics, California State University, Fresno CA, United States of America
- <sup>g</sup> Also at Faculty of Physics and Applied Computer Science, AGH-University of Science and Technology, Krakow, Poland
- <sup>*h*</sup> Also at Department of Physics, University of Coimbra, Coimbra, Portugal
- <sup>*i*</sup> Also at Università di Napoli Parthenope, Napoli, Italy
- <sup>j</sup> Also at Institute of Particle Physics (IPP), Canada
- <sup>k</sup> Also at Department of Physics, Middle East Technical University, Ankara, Turkey
- <sup>1</sup> Also at Louisiana Tech University, Ruston LA, United States of America
- <sup>m</sup> Also at Group of Particle Physics, University of Montreal, Montreal QC, Canada
- <sup>n</sup> Also at Institute of Physics, Azerbaijan Academy of Sciences, Baku, Azerbaijan
- <sup>o</sup> Also at Institut für Experimentalphysik, Universität Hamburg, Hamburg, Germany
- <sup>p</sup> Also at Manhattan College, New York NY, United States of America
- <sup>q</sup> Also at School of Physics and Engineering, Sun Yat-sen University, Guanzhou, China
- <sup>r</sup> Also at Academia Sinica Grid Computing, Institute of Physics, Academia Sinica, Taipei, Taiwan
- <sup>s</sup> Also at High Energy Physics Group, Shandong University, Shandong, China
- <sup>t</sup> Also at Section de Physique, Université de Genève, Geneva, Switzerland
- <sup>*u*</sup> Also at Departamento de Fisica, Universidade de Minho, Braga, Portugal
- <sup>v</sup> Also at Department of Physics and Astronomy, University of South Carolina, Columbia SC, United States of America
- <sup>w</sup> Also at KFKI Research Institute for Particle and Nuclear Physics, Budapest, Hungary
- <sup>x</sup> Also at California Institute of Technology, Pasadena CA, United States of America
- <sup>y</sup> Also at Institute of Physics, Jagiellonian University, Krakow, Poland
- <sup>z</sup> Also at Department of Physics, Oxford University, Oxford, United Kingdom
- aa Also at Institute of Physics, Academia Sinica, Taipei, Taiwan
- <sup>ab</sup> Also at Department of Physics, The University of Michigan, Ann Arbor MI, United States of America
- *ac* Also at DSM/IRFU (Institut de Recherches sur les Lois Fondamentales de l'Univers), CEA Saclay (Commissariat a l'Energie Atomique), Gif-sur-Yvette, France
- <sup>ad</sup> Also at Laboratoire de Physique Nucléaire et de Hautes Energies, UPMC and Université Paris-Diderot and CNRS/IN2P3, Paris, France
- ae Also at Department of Physics, Nanjing University, Jiangsu, China

\* Deceased